Documents de travail

N° G2020/13 Décembre 2020

Market Power and Labor Share

Arthur BAUER - Jocelyn BOUSSARD





Institut national de la statistique et des études économiques

G2020/13

Market Power and Labor Share

Arthur BAUER^{*} Jocelyn BOUSSARD^{**}

Décembre 2020

Département des Études Économiques – Timbre G201 88, avenue Verdier – CS 70 058 – 92 541 MONTROUGE CEDEX – France Tél. : 33 (1) 87 69 59 54 – E-mail : <u>d3e-dg@insee.fr –</u> Site Web Insee : <u>http://www.insee.fr</u>

Ces documents de travail ne reflètent pas la position de l'Insee et n'engagent que leurs auteurs. Working papers do not reflect the position of INSEE but only their author's views.

^{*} INSEE – Crest. Email : <u>arthur.bauer@ensae.fr</u>

^{**} Commission Européenne – Crest. Email: <u>boussard.jocelyn@ensae.fr</u>. Website : <u>https://jboussard.com</u>

Nous remercions les départements des études économiques, et des archives et de la statistique d'entreprise de l'INSEE, notamment Kim Anh Fabre, Pascal Lucas, Christelle Le Borgne et Emmanuel Gros qui nous ont permis d'accéder aux données et de mieux les comprendre. Certaines parties de la présente étude ont été menées alors que Jocelyn Boussard était économiste au sein du Service d'étude des politiques structurelles de la Banque de France. Nous remercions Elizaveta Archanskaia, Sabien Dobbelaere, Dominique Goux, Maxime Gueuder, Francis Kramarz, Edouard Jousselin, Rémy Lecat, Margarita Lopez-Forero, Jacques Mairesse, Rémi Monin, Ludovic Panon, Sébastien Roux, Olivier Simon et tous les participants aux séminaires de l'INSEE, de la Banque de France, de la DG-ECFIN, du Collège de France et du CREST qui nous ont fait part de leurs commentaires utiles.

Pouvoir de marché et part du travail

Les tendances séculaires du pouvoir de marché et de la part du travail dans la valeur ajoutée ont des conséquences importantes sur les inégalités et l'efficacité de l'allocation des facteurs de production. Pour les analyser, il faut recueillir des données exhaustives et renseignant le détail du compte de chaque entreprise, couvrant plusieurs dizaines d'années. Nous exploitons une nouvelle base de données sur l'univers des entreprises françaises entre 1984 et 2016 et documentons une hausse de la concentration depuis le début des années 1990. Malgré une certaine stabilité de la part agrégée du travail, les entreprises de plus grande taille dont la part du travail est moins importante ont gagné des parts de marché, surtout dans les industries où la concentration a le plus augmenté. Le taux de marge, i.e. le ratio du prix au coût marginal, d'une entreprise *type* – que nous considérons ici comme un indicateur de son pouvoir de marché – a diminué, mais la redistribution des parts de marché vers les entreprises de plus grande taille a fait augmenter le taux de marge agrégé. Notamment, selon nos constatations, la hausse de la concentration ne s'accompagne pas d'un accroissement du pouvoir de marché dans les tranches supérieures. Nous montrons également qu'il est essentiel de tenir compte de la redistribution dans l'ensemble des entreprises pour bien comprendre comment les tendances du pouvoir de marché ont façonné la dynamique de la part globale du travail en France.

Mots-clés : Part du travail, taux de marge, concurrence, fonction de production

Market Power and Labor Share

Secular trends in market power and labor share have important implications for inequality and allocative efficiency. Studying them requires comprehensive and detailed firm-level data spanning several decades. For that purpose, we leverage a novel and detailed database on the universe of French firms between 1984 and 2016, that we use to document a rise in concentration in France since the beginning of the 1990s. Despite a relative stability of the aggregate labor share, we show that firms with lower labor shares have been gaining market shares. As low labor share firms also tend to be larger, this market share reallocation has been stronger in industries where concentration increased the most. We rely on markups as proxies of firm-level market power, and on a flexible production function that allows the identification of firm-specific output elasticities and markups. We find that the markup of the *typical* firm has decreased, but the reallocation of market shares toward larger firms contributed to an increase of the aggregate markup. Finally, we show how taking into account reallocation across firms is essential to understand how the aggregate market power evolution has shaped the dynamics of the aggregate labor share in France.

Keywords: Labor share, markup, competition, production function

Classification JEL: E10, E23, E25

1 Introduction

Large and productive superstar firms have been gaining market shares in many advanced economies, and the rise of their market power, measured either through their markup or their profitability, has been the focus of attention in many recent works. De Loecker et al. (2020) have documented an increase in top firm market power in the US that is large enough to have important macroeconomic consequences. They find that the weighted average markup in the United States rose from 21% above marginal cost at the beginning of the 1980s to around 61% now. Autor et al. (2020) also document a rise of the weighted average markup in the US. Gutiérrez and Philippon (2018) argue that European markets are more competitive, and exhibit lower levels of concentration, lower excess profits and lower barriers to entry, which raises the question of whether the secular trends above are specific to the US. We use detailed firm-level administrative data on the universe of French firms to document facts about market power and labor shares in France.

These questions are important for inequality concerns. One of the important macroeconomic implications of a rise of market power is a decline in the aggregate share of income going to workers. Given that there is ample evidence that labor is more evenly distributed than capital (Garbinti et al., 2018; Piketty et al., 2018) or firm ownership (Bauer et al., 2018), a decline in the aggregate labor share is a possible driver of inequality. Important work has shown that the aggregate labor share has indeed been declining in a wide range of countries (Karabarbounis and Neiman, 2014; Elsby et al., 2013; Grossman et al., 2018). Using aggregate data, Barkai (2020) and Boussard and Lee (2020) show that both the labor share and capital shares have declined in the United States and many advanced economies, while measures of the profit share have increased. Looking more closely at firm-level data, Autor et al. (2020); Kehrig and Vincent (2018) show that the labor share of the typical firm has actually increased, while the aggregate fall is attributable to reallocation from high to low-labor share firms.

Market power trends have also important but ambiguous consequences for allocative efficiency. As shown by Baqaee and Farhi (2020), a reallocation of market shares to high-markup firms, as in Autor et al. (2020) increases efficiency, but an increase in markup dispersion, as in De Loecker et al. (2020) reduces efficiency. Market power also has important but ambiguous dynamic implications: while lower competition may lead firms to under-invest (Gutiérrez and Philippon, 2017), the relationship between competition and innovation depends on the initial level of competition (Aghion et al., 2005).

Understanding the underlying micro-structural transformations behind these aggregate trends is crucial to identify their possible explanations such as changes in the competitive environment and changes in technology. For instance, Bonfiglioli et al. (2019) and Panon (2020) show that national firms compete in markets that are increasingly global, which reduces firm-level markups but benefits larger firms, and Melitz (2003) Mayer et al. (2014) show that international competition causes reallocation toward top producers. Recent work (Autor et al., 2020; Van Reenen, 2018) argues that technological change, such as the growth of platform competition in digital markets, may have caused reallocation from small to large firms that could lead to dominance by a small number of firms. Lashkari et al. (2019) find that the rise of Information Technology has disproportionately benefited larger firms.

We use France as a laboratory to study the link between variations in industry concentration and firm-level outcomes, and provide evidence on the sources of market power variations. France is an interesting case because in contrast to the US, the labor share in France appears to have been stable or increasing over the past decades (see Figure 1).

We document important facts about secular trends in France that are similar to what has been documented for other advanced economies. When we decompose labor share variations in France, we show that there has been an important reallocation of market shares from firms with high labor shares to firms with low labor shares, which tend to be larger. This reallocation is correlated with a rise of industry concentration, measured through a wide range of proxies, from the beginning of the 1990s. However, labor shares have on average increased at all points of distribution, a development that has offset the effect of reallocation and explains why the aggregate labor share in France is broadly stable.

To assess the extent to which firm-level market power dynamics has played a role in explaining the divergence between firm-level labor share in France and the US, as opposed to other explanations like technological change, we estimate firm-level markups and output elasticities using a flexible production function that allows variations in the marginal product of inputs both across firms and time periods. We follow De Loecker and Warzynski (2012) and first estimate firm-level elasticities of value-added to labor and capital, and then recover markups by assuming that firms minimize their costs and that labor is a flexible input. We rely on unique and comprehensive administrative data covering the universe of French firms. This data is produced by the French National Institute of Economics and Statistics (INSEE) and contains standard income and balance sheet information for almost all firms in France from 1984 to 2016.

Importantly, we find no evidence that the rise in concentration translated into an increase in firmlevel market power. We find that there is substantial heterogeneity in markups, and that markups are increasing with firm size. We also find that much of the increase in firm-level labor shares is attributable to decreases in firm-level markups. All in all, high-markup firms gained market shares while the markup of the typical firm decreased, which indicates both an improvement in allocative efficiency and a reduction of the distorsive effect of markups. We also show that this reallocation is strongly correlated with the rise in concentration at the industry level.

Our paper contributes to the macroeconomic literature that documents a number of important secular trends that have recently swept across advanced economies. A number of recent papers have documented growing industry concentration and within-industry dispersion in firm outcomes (Andrews et al., 2016; Berlingieri et al., 2017; Song et al., 2018; Card et al., 2013). In parallel, there is a large body of evidence on a global fall in the labor share across many industries (Elsby et al.,

2013; Karabarbounis and Neiman, 2014, 2018; Barkai, 2020; Boussard and Lee, 2020; Grossman et al., 2018). We show that concentration and firm-level market power are not necessarily correlated, even at the top, even though at the aggregate level the reallocation of market shares toward high-markup firms contributes to a rise in the aggregate markup. Our results (i) that firm-level markups have decreased, and (ii) that reallocation towards high markup firms reflects a rise in concentration and contributes to a rise in aggregate markup are consistent with Autor et al. (2020). However, in France, the decrease in firm-level markups is larger, and the reallocation effect does not offset it.¹ This difference is also consistent with evidence in Gutiérrez and Philippon (2018) that European markets have become more competitive than US markets.

The rest of the paper is organized as follows. Section 2 presents our theoretical framework, Section 3 presents our strategy for estimating firm-level markups, Section 4 presents the data that we use to implement this strategy, Section 5 documents important macro and micro facts about labor share and concentration in France, and Section 6 presents our results about markups in France. Finally, Section 7 concludes.

2 Theoretical Framework

In this section, we provide a general theoretical framework that allows us to map variations in aggregate the labor share to variations in firm-level market power, input elasticities and market shares. Consider an industry with N firms indexed by *i*. Consistent with a wealth of evidence and in the spirit of canonical models (Melitz, 2003; Hopenhayn, 1992), we assume that firms have heterogeneous exogenous productivity Ω_{it} and have access to a common production technology $\mathcal{Q}(.)$ defined as:

$$Y_{it} = \mathcal{Q}(\Omega_{it}, L_{it}, K_{it}),$$

that they use to produce value-added Y_{it} , using variable labor input L_{it} , and capital stock K_{it} . We assume that adjusting the capital stock is subject to cost $C_a(.)$, which depends only on the current and previous levels of capital, and crucially not on variable inputs levels. The sum of discounted costs of the firm is:

$$\mathcal{V}(\boldsymbol{Z}_{it}) = \min_{\boldsymbol{X}_{it}} \mathcal{C}(\boldsymbol{X}_{it}, \boldsymbol{Z}_{it}) + \beta \mathbb{E} \left[\mathcal{V}(\boldsymbol{Z}_{it+1}) \right],$$

s.t $\mathcal{Q}(\Omega_{it}, \boldsymbol{X}_{it}) = Y_{it},$

where C(.) is the total cost of the firm, $X_{it} = (L_{it}, K_{it})$ refers to inputs, and Z_{it} to variables that are exogenous to the firm at time t, such as previous year capital stock, productivity and input

¹Possible interpretations of these difference are that the market power of French firms is more sensitive to the underlying cause, for instance if French firms are more exposed to globalization or to competition on internet platforms than US firms, or if the productivity gap between top French firms and laggards is not as large as for top US firms.

prices. The Lagrangian associated with the right-hand-side of the Bellman equation is defined as:

$$\mathcal{L}(\boldsymbol{X}_{it}, \xi_{it}, Y_{it}, \boldsymbol{Z}_{it}) = W_{it}L_{it} + r_{it}\left(K_{it} + \mathcal{C}_a(K_{it}, K_{it-1})\right) + F_{it} + \beta \mathbb{E}\left[\mathcal{V}(\boldsymbol{Z}_{it+1})\right] - \xi_{it}(\mathcal{Q}(\Omega_{it}, \boldsymbol{X}_{it}) - Y_{it}),$$

where W_{it} is the wage, r_{it} is the user cost of capital, F_{it} is an exogenous fixed cost, and ξ_{it} is the Lagrange multiplier. The first-order conditions at the optimal choice of inputs X_{it}^* and ξ_{it}^* imply that:

$$\nabla \mathcal{L}(\boldsymbol{X}_{it}^*, \xi_{it}^*, Y_{it}, \boldsymbol{Z}_{it}) = 0, \tag{1}$$

where ∇ denotes the gradient vector of partial derivatives with respect to inputs. Applying Equation (1) to the flexible labor input yields the following cost-minimization condition linking the wage and marginal product of labor:

$$\frac{\partial \mathcal{L}}{\partial L}(\boldsymbol{X}_{it}^{*}, \xi_{it}^{*}, Y_{it}, \boldsymbol{Z}_{it}) = W_{it} - \xi_{it}^{*} \frac{\partial \mathcal{Q}}{\partial L}(\Omega_{it}, \boldsymbol{X}_{it}^{*}, .) = 0.$$

The output elasticity with respect to the labor input L, $\theta_{l,it}$, can therefore be expressed at the optimum as:

$$\theta_{l,it} \equiv \frac{L_{it}^*}{Y_{it}} \frac{\partial \mathcal{Q}}{\partial L}(\Omega_{it}, \boldsymbol{X}_{it}^*) = \frac{1}{\xi_{it}^*} \frac{W_{it} L_{it}^*}{Y_{it}}.$$
(2)

Using the first order conditions in Equation (1) to express the optimal choice of inputs X_{it}^* and ξ_{it}^* as functions of output Y_{it} and exogenous variables Z_{it} , we derive the optimal total cost as a function of output and exogenous variables:

$$\mathcal{C}^*(Y_{it}, \boldsymbol{Z}_{it}) = \mathcal{C}(\boldsymbol{X}_{it}^*(Y_{it}, \boldsymbol{Z}_{it}), \boldsymbol{Z}_{it}),$$

At the optimum, the Lagrangian is equal to total cost, and from the envelop theorem it follows that the marginal cost is equal to the Lagrange multiplier ξ_{it}^* :

$$\frac{\partial \mathcal{C}^*}{\partial Y}(Y_{it}, \boldsymbol{Z}_{it}) = \frac{\partial \mathcal{L}^*}{\partial Y}(Y_{it}, \boldsymbol{Z}_{it}) = \frac{\partial \mathcal{L}}{\partial Y}(\boldsymbol{X}^*_{it}, \xi^*_{it}, Y_{it}, \boldsymbol{Z}_{it}) = \xi^*_{it}$$

Dropping for simplicity the superscript * to denote optimal variables, we define the markup as the ratio of the output price of the firm P_{it} to the marginal cost:

$$\mu_{it} = \frac{P_{it}}{\xi_{it}}.$$
(3)

The markup captures the degree of pricing power of the firm, and is a widely used measure of firmlevel market power. As noted by De Loecker and Warzynski (2012), this expression is robust to various static price setting models, and does not depend on any particular form of price competition among firms. The markup itself will, however, depend on the specific nature of competition among firms. Moreover, it follows from Equations (2)) and (3) that the markup is defined as the elasticity of output with respect to the labor input, divided by the share of this labor costs in total firm revenue, *i.e* the labor share λ_{it} :

$$\mu_{it} = \theta_{l,it} \frac{P_{it} Y_{it}}{W_{it} L_{it}} \equiv \frac{\theta_{l,it}}{\lambda_{it}}.$$
(4)

It is important to note that Equation (4) only applies to inputs that are freely adjustable, at least at the margin. In the case of capital, the relationship between the markup and the output elasticity becomes:

$$\mu_{it} = \frac{\theta_{k,it}}{\lambda_{it}^k (1 + \Delta_{a,it})},\tag{5}$$

where λ_{it}^k is the capital share of revenue, $\theta_{k,it}$ is the output elasticity with respect to capital, and $\Delta_{a,it} = \frac{\partial C_a}{\partial K}(K_{it}, K_{it-1}) + \frac{\partial C_a}{\partial K_{-1}}(K_{it+1}, K_{it})$ is the wedge attributable to the adjustment costs. The sign of the wedge is not straightforward and depends on the convexity of the adjustment cost function as well as on expectations of future target stock of capital. De Loecker and Warzynski (2012) show that abstracting from adjustment costs generally results in a negative wedge, and therefore an overestimated markup.²

Another important source of gap between the output elasticity of an input and its share in revenue is when firms are not price-takers on the market for inputs, for instance if the firm has monopsony power in the labor market, or engages in efficient bargaining (Dobbelaere and Mairesse, 2013; Dobbelaere and Kiyota, 2018). In that case, the relationship between the markup and the output elasticity becomes:

$$\mu_{it} = \frac{\theta_{l,it}}{\lambda_{it}(1 + \Delta_{m,it})},\tag{6}$$

where the sign of the wedge $\Delta_{m,it}$ depends on the labor market setting. Dobbelaere and Kiyota (2018) show that in the case of efficient bargaining, the wedge is positive, and in the case of monopsony the wedge is negative. We abstract from these two possible wedges but discuss in more detail the implications for our results in section 6.

In what follows, we map the aggregate labor share into firm level markups, and the output elasticity of labor. First, we define the aggregate labor share Λ_t as the value-added-weighted average of firm-level labor shares:

$$\Lambda_t \equiv \frac{\sum_i W_{it} L_{it}}{\sum_i P_{it} Y_{it}} = \sum_i S_{it} \lambda_{it},$$

where $S_{it} = \frac{P_{it}Y_{it}}{\sum_{i}P_{it}Y_{it}}$ is the market share of firm *i*. From Equation (4) we know that the labor share is the product of the output elasticity of labor and the inverse markup:

$$\lambda_{it} = \theta_{l,it} \mu_{it}^{-1}.\tag{7}$$

We decompose the output elasticity of labor $\theta_{l,it}$ into a component stemming from returns to scale, which tells us how much output expands when all inputs increase proportionally, and a component

 $^{^{2}}$ See Doraszelski and Jaumandreu (2019) for a discussion of the sign of adjustment cost wedges.

stemming from the labor intensity of the production process relative to capital:

$$\theta_{l,it} = \underbrace{\theta_{l,it} / (\theta_{l,it} + \theta_{k,it})}_{\text{Labor Intensity}} \underbrace{(\theta_{l,it} + \theta_{k,it})}_{\text{Returns to Scale}} \equiv \alpha_{it} \gamma_{it}, \tag{8}$$

noting that when α_{it} is high the production process is intensive in labor relative to capital. It follows from Equations (2), (7), and (8) that the aggregate labor share can be expressed as a function of firm level labor intensity, returns to scale, and markups:

$$\Lambda_t = \sum_i S_{it} \alpha_{it} \gamma_{it} \mu_{it}^{-1}.$$
(9)

We compute the aggregate markup M_t as the value-added weighted harmonic average of firm-level markups:

$$M_{t} \equiv \left[\frac{\sum_{i} P_{it} Y_{it} \mu_{it}^{-1}}{\sum_{i} P_{it} Y_{it}}\right]^{-1} = \left[\sum_{i} S_{it} \mu_{it}^{-1}\right]^{-1},$$

In the next section, we describe the estimation procedure we follow to recover estimates of firmlevel output elasticities of labor and capital, which together with firm-level labor and market shares observed in the data, allows us to compute the contribution of markups, labor intensity, and returns to scale to the aggregate labor share.³

3 Estimation Procedure

To recover markup from production data, we rely on Equation (4). This framework is particularly convenient to analyze the evolution of markups in the long run because it does not require observing consumer-level attributes to estimate demand elasticities. Second, it makes no assumption on firms pricing behavior and competition environment. It only requires two assumptions: firms minimize production cost and freely adjust at least one variable input.

We can directly observe firm-specific input shares in production data. It is not the case for output elasticities with respect to inputs. Because these elasticities can vary across time and firms, we estimate a flexible production function, with a minimum number of parametric restrictions. In what follows, we assume that firms belonging to a particular industry j share the same technology $f_j(.)$, using labor and capital to generate value added. Moreover, we assume that productivity is Hicks-neutral and evolves according to an AR(1) Markov process. For firm i in industry j, our empirical model is given by:

$$\int y_{it} = f_j(k_{it}, l_{it}) + \omega_{it} + \epsilon_{it}, \qquad (10)$$

³We abstract from input-output linkages by considering value-added production function. Baqaee and Farhi (2020) show that input-output linkages are important for the propagation of productivity shocks, and Grassi (2017) shows that they matter for market power in the case oligopolistic competition.

where y_{it} stands for the logarithm of value added firm *i* at time *t*, and l_{it} and k_{it} are the logarithms of employment and capital stock. Productivity ω_{it} is Hicks-neutral, ϵ_{it} is an i.i.d measurement error, and ξ_{it} is the i.i.d innovation to productivity. Steady-state productivity η_{jt} and time trend ν_{jt} are common across firms in industry *j* in period *t*.

One issue that prevents us for simply running Ordinary Least Squares (OLS) on Equation (10) is that we do not observe productivity ω_{it} but firms have information about their productivity when they choose their inputs. ω_{it} is therefore correlated with $k_i t$ and $l_i t$ and OLS estimates are biased. In what follows, we make the following standard assumptions regarding the timing of firm decisions:

Assumption 1. (Information Set) The firm's information set at t, i.e. I_t , includes current and past productivity shocks $\{\omega_{i\tau}\}_{\tau=0}^t$ but does not include future productivity shocks $\{\omega_{i\tau}\}_{\tau=t+1}^\infty$. Measurement errors ϵ_{it} satisfy $\mathbb{E}[\epsilon_{it}|I_t] = 0$. The productivity process defined in Equation eq11 is known to firms and stochastically increasing in ω_{it-1}

Assumption 2. (Input Choices) Labor and capital inputs used at time t are chosen with information set I_t .

Assumptions (1) and (2) are straightforward: firms do not observe ω_{it} until time t, but the Markov process defines what the firm knows about the distribution of future productivity shocks. We rely on an approach to control for unobserved productivity that is usually called dynamic panel estimation (Blundell and Bond, 2000). We use the AR(1) structure of the productivity process to write current value-added as :

$$y_{it} = \rho_{jt}y_{it-1} + (f_j(k_{it}, l_{it}) - \rho_{jt}f_j(k_{it-1}, l_{it-1})) + \eta_{jt} - \nu_{jt}t + u_{it},$$

where the composite error $u_{it} = \xi_{it} + \epsilon_{it} - \rho \epsilon_{it-1}$ has by assumptions (1) and (2) a zero mean conditional on information set I_{t-1} . Conditioning on a set of instruments included in I_{t-1} , we estimate the model using non-linear GMM. Our moment conditions can be written as:

$$\mathbb{E}\left[u_{it}|I_{t-1}\right] = \mathbb{E}\left[y_{it} - \rho_{jt}y_{it-1} - \left(f_j(k_{it}, l_{it}) - \rho_{jt}f_j(k_{it-1}, l_{it-1})\right) - \eta_{jt} - \nu_{jt}t|I_{t-1}\right] = 0.$$
(12)

We assume that technology $f_j(.)$ in sector j is a translog production function of capital and labor:

$$f(k_t, l_t) = \beta_{l,jt} l_{it} + \beta_{k,jt} k_{it} + \beta_{ll,jt} l_{it}^2 + \beta_{kk,jt} k_{it}^2 + \beta_{lk,jt} l_{it} k_{it},$$

and we use past values ω_{it-1} , l_{it-1} , m_{it-1} , k_{it-1} and higher order combinations of those terms, a time trend t and a constant as instruments in Equation (12). From the estimates of the parameters of the production function, we compute the firm-level output elasticity of labor and capital for firm i in year t as:

$$\theta_{l,it} = \beta_{l,jt} + 2\beta_{ll,jt}l_{it} + \beta_{lk,jt}k_{it},$$

$$\theta_{k,it} = \beta_{k,jt} + 2\beta_{kk,jt}k_{it} + \beta_{lk,jt}l_{it}.$$

From Equation (8), we retrieve firm-level labor intensity and returns to scale.

Previous studies estimating markups with production data have often estimated production functions with the proxy variable method. This method relies on a non-parametric estimation of unobserved productivity ω_{it} from observed variables using the assumption that some proxy variable, either investment (Olley and Pakes, 1996) or intermediate input demand (Levinsohn et al., 2003; Ackerberg et al., 2015), is an invertible function only of other inputs and productivity. However, this approach is not valid if the proxy variable is also a function of some unobserved shock, such as an input cost shock to all inputs, or a demand shock. Let us define intermediate input demand m_{it} as a function of capital, labor, productivity, and some unobserved shock d_{it} :

$$m_{it} = m(\omega_{it}, k_{it}, l_{it}, d_{it}). \tag{13}$$

Assuming that this function is invertible in ω_{it} and using Equation (10), one can write that value added y_{it} is an unknown function of inputs and the unobserved shock:

$$y_{it} = f_j(k_{it}, l_{it}) + \omega(m_{it}, k_{it}, l_{it}, d_{it}) + \epsilon_{1,it} = g(m_{it}, k_{it}, l_{it}, d_{it}) + \epsilon_{1,it}.$$
(14)

Ignoring the unobserved shock, and using assumption (1) that ϵ_{it} is independent from input choices, we can apply Ackerberg et al. (2015) and obtain a non parametric estimate \hat{g}_{it} of g(.) that is a high-order polynomial in m_{it} , k_{it} , and l_{it} , but not of d_{it} :

$$y_{it} = \hat{g}_{it} + \hat{\epsilon}_{it},\tag{15}$$

where the residuals $\hat{\epsilon}_{it}$ are correlated with d_{it} . In practice, when we apply this procedure, we find that the residuals are not i.i.d. As Doraszelski and Jaumandreu (2019) have recently discussed, d_{it} , as ω_{it} , should also be recognized as an autocorrelated unobservable. If so, the instruments used in the second stage of the proxy variable method are not consistent. Nevertheless, because the proxy variable method has been widely used to study aggregate markup trends, we present in appendix results obtained following this method.

4 Data

To carry out our empirical analysis we rely on several sources of micro data produced by the French Institute of Statistics and Economic Studies (INSEE), covering the universe of French firms spanning the 1984-2016 period. These data are in particular one of the main sources of the elaboration of National Accounts. Our sources are gathered out of the universe of firms' tax forms and provide balance sheet, income, and cost information at the firm level, as well as employment, the industry in which the firm operates, the type of legal entity (micro-firms, sole proprietorship entities, or limited liability companies and corporations) and the tax regime to which it is affiliated (micro-regime, simplified regime, or normal regime). From 1984 to 2007, we rely on the SUSE sources (Système Unifié de Statistiques d'Entreprises), gathering information from firms affiliated to two tax regimes, the BRN regime (Bénéfice Réel Normal) and RSI regime (Régime Simplifié d'Imposition). These files allow to distinguish between payments to labor, material inputs, other intermediary inputs, and investment, and provide information of the book value of capital of the firm and total employment. Hence, they have been widely used in previous research (di Giovanni et al., 2014; Caliendo et al., 2015).

From 2008, we rely on the ESANE sources (*Élaboration des Statistiques Annuelles d'Entreprises*), that result from the unification of the previous SUSE data with Annual Surveys of Firms that were conducted each year for broad sectors of industries. Because there is some overlap of information between tax forms and surveys, INSEE has a process to reconcile diverging information. To construct our panel of firms we exclude from the post-2008 data firms affiliated to the micro-BIC regime.⁴ Moroever, we restrict our analysis to legal units with a unique and valid identifier number.⁵

We focus on market sectors⁶ and exclude agriculture because our sample does not cover well firms in that sector, which are mostly affiliated to a tax regime that is not included in the micro-BIC, BRN and RSI regimes. We also exclude real estate and finance, because we focus on the production side of value-added distribution among workers and owners of capital and firms. We have 5.7 million firms in our sample, 3.7 million of which have at least one employee. Finally, we rely on industry-level data from KLEMS (van Ark and Jäger, 2017) for information on investment and output prices to compute deflated values for value-added and capital stocks.

Overview of the data

Table 1 reports the year-by-year total number of observations, as well as aggregate labor costs, value-added, investment, both in level and in share of their aggregate values for the corporate sector in France. There are on average 800 thousand observations per year, accounting for 87% of total labor costs, 84% of total value-added, with little variations over time. Our data only accounts for 68% of total investment in the corporate sector. This is due to the fact that many small firms affiliated to the simplified regime report missing investment. To construct measures of capital input, we use instead the reported book values of the capital stock.

Table 2 describes the main variables that we use in our empirical analysis. Our sample of 3.7 million firms with at least one employee spans over 33 years, with 27 millions firm-year observations, average sales is 2.6 M \in , average number of employees: 14, and average capital stock: 1.3 M \in . This data is highly skewed as the median level of sales is 285 K \in , median number of employees is 3, and median

⁴An extremely simplified regime introduced in 2008 applicable to very small firms, whose total sales do not exceed 170 K \in if the firm operates within the real estate and trade sectors, or 70 K \in otherwise. This regime has been widely used by free-lance workers who do not report any capital nor employment.

 $^{{}^{5}}$ A firm is defined as a legal unit with a unique SIREN identifying number. In ESANE, legal units belonging to the same conglomerate are brought together and their accounts are consolidated (Deroyon, 2015). We do not use this information here.

⁶The market sectors are total economy excluding public administrations, healthcare, and education.

capital stock: 76 K \in . This reflects the fact that our data is nearly exhaustive and includes many small firms. For firms that report non missing investment, the average reported value is 185 K \in , and the median is 4 K \in , which also partly reflects the fact that investment is lumpy.⁷ The average labor share in our sample, computed as the ratio of the sum of the wage bill and payroll taxes to value-added, is 75%, close to the median at 74%.

Aggregate Labor Share

Figure 1 reports the ratio of compensation of employees, including payroll taxes, to total valueadded in the macro and micro data, from 1984 to 2016. The aggregate labor share in our sample, is lower than the average firm-level labor share. As discussed below in Section 5, larger firms have a lower labor share, which brings down the weighted average labor share. In the sample of firms with at least one employee on which we rely in the rest of the paper, the aggregate labor share decreases from 69.3% in 1984 to 64.7% in 2000, and then increases back to a level close to its initial level, reaching 69.1% in 2016. The aggregate level is on average 67.1% over the period. Aggregate data in principle also includes firms that have no employee, and doing so in our micro data decreases the aggregate level of the labor share by around 1 percentage point: it stands at 66.1% of value-added on average over the period, and has the same U-shaped trajectory. This aggregate pattern differs substantially from the decrease of the labor share in the US, discussed by Autor et al. (2020); Kehrig and Vincent (2018), while others have argued that France, as many advanced economies, also experienced a secular decrease in the labor share (see e.g Grossman et al., 2018; Karabarbounis and Neiman, 2014). Because of the U-shaped trajectory of the labor share, both in the micro and macro data, we find that conclusions of a secular decline in France are misguided.

Our sample is limited to incorporated firms in the market economy outside agriculture, real estate, and finance. Despite the fact that there is no available aggregate data for France for this particular sample, the aggregate labor share in our data closely matches the aggregate patterns of the labor share that we can measure on similar spheres of activity, both in levels and in variations.

French National Accounts provide detailed operating accounts for spheres that are larger than our data in various dimensions. Figure 1 reports the labor share of the entire corporate sector, including corporations operating in the agriculture, real estate, and finance. Before 2000, the average level of the labor share in the corporate sector, reported by INSEE, is the same as the aggregate labor share in our sample including firms with no employees (65.4%). It starts from a slightly higher level in 1984 (71.6%) than our sample estimate (68.4%) and reaches a slightly lower level in 2000 (63.4% as opposed to 64.1% in our sample). After 2000, however, the corporate labor share rises by 2 percentage points, but the labor share in our sample rises by 4 percentage points.

Figure 1 also reports the total labor share (corporate and non-corporate) excluding agriculture, real estate, and finance. The non-corporate sector is mainly composed of self-employed workers with few salaried workers. As a result, the total labor share reported by INSEE is lower - on average 61%

⁷Average mean firm investment across years is 140 K€ and mean investment is 8 K€.

over the period, against 66.1% in our data with all firms. Nevertheless, after 2000, and despite this difference in levels, the rise of the total labor share measured with the same industry composition as our data matches the 4 percentage point increase that we observe in our data. One possible explanation, as Cette et al. (2019) discuss, is that because the real estate sector has a labor share close to zero, its growing share in total value-added contributes negatively to the aggregate labor share of the corporate sector, especially during the housing boom after 2000.

5 Labor Share and Concentration

In this section, we revisit five important facts about concentration and labor shares in the French context. In particular, we find that the rise in concentration in France is associated with an increase in firm-level labor shares, and a reallocation of market shares towards large and low-labor-share firms.

Rise in concentration

Figure 2 reports the cumulative change since 1984 in sales weighted average levels of industry concentration indexes, where each index measures concentration of sales at the 3-digit national industry level. The share of sales of the largest 1% or 5% firms in each industry increased sharply on average since 1984, by 9 and 7 percentage points respectively. The concentration ratios, defined as shares of the 4 and 20 largest firms in each industry, followed a different pattern before 1995, but have increased by close to 4 percentage points each on average since 1995.⁸

Overall, we find that concentration ratios and top shares have increased in more than half of the 211 industries since 1995: the median increase of both concentration ratios is 2 percentage points, and the median increases of the top 1% and 5% shares are 4 and 5 percentage points respectively. Figure E.1 in the Appendix shows that the average rise in concentration is observed across broad sectors of the economy: the magnitude of the increase is similar in both the manufacturing (3 to 7 percentage points from the lowest point to 2016, depending on the index) and non-manufacturing sector (4 to 10 percentage points).

These results are consistent with evidence across the US and other OECD countries (CEA, 2016; Autor et al., 2020; Andrews et al., 2016). It is important to note that concentration is calculated as the share of sales of largest firms in a narrowly defined industry at the national level. It is best to interpret it as a measure of firm dispersion of outcomes rather than firm market power, as the national industry level is not necessarily the relevant market. In fact, Section 6 shows that the industry-level correlation between these measures of concentration and the largest firms market power is not significantly positive.

⁸The median 3-digit industry has around 900 firms in a given year, but because 25% of the industries have more than 5,000 firms, and 25% have less than 200 firm, the number of firms in the top 1% and 5% differs greatly from one industry to the next. The median size of the 3-digit manufacturing industry is around 500 and the median size of the 3-digit non manufacturing industry is 3,600.

Reallocation of labor shares

We build on Kehrig and Vincent (2018) and decompose the variations of the aggregate labor share to understand whether they are driven by variations at the firm level or by composition effects. Figure 3 reports for each decile of labor share, the value-added-weighted average labor share and the share of industry value added of firms in that decile, in the first and last five years of the sample. To account for industry-specific differences in the joint distributions of labor share and value-added, they are first calculated within each 3-digit industry. These distributions are then averaged across these 211 industries using value-added weights in a given period. The vertical bars illustrate how low labor share firms gained market share in the last 30 years. Firms in the lowest decile of labor share accounted for 12% of their industry value-added before 1990, compared to 16% in after 2010. The rise in industry shares is verified for four out of the five lowest deciles of labor share, while all five highest deciles of labor share accounted for less of industry value-added in 2011-2016 than in 1984-1989. The connected lines illustrate how the raw distribution of labor shares has shifted upwards. The average labor share of each decile is higher in after 2010 than before 1990. Figure E.2 in the Appendix shows that these patterns are observed across broad sectors of the economy, in manufacturing as well as non-manufacturing industries.

Figure 4 quantifies the contribution of three components to the variation of the aggregate labor share in France: a term accounting for reallocation of value added shares across industries, and the two within-industry components discussed above: within and across quantiles of labor shares.⁹ The figure first confirms that reallocation across industries plays only a minor role for aggregate labor share variations. The increase in the value-added shares of low-labor-share firms contributed to an accumulated 5 percentage points decrease of the aggregate labor share since 1984, holding the distribution of labor shares constant from one year to the next. This was offset by the upward shift in the labor share distribution, that contributed to a rise of the aggregate labor share of 5 percentage points, holding the value-added distribution constant.

Figure E.3 in the Appendix presents the 1984-2016 cumulative results of the same decomposition for manufacturing and non-manufacturing industries separately. In both macro sectors, as in the whole economy, the reallocation across industries had a negligible impact, the reallocation of value-added contributed negatively to the aggregate labor share and the labor share distribution shifted upwards and contributed positively to the aggregate labor share. In manufacturing, the aggregate labor share decreased because the upward shift in the labor share distribution did not offset the reallocation.¹⁰

Our decompositions show that the typical industry in France experienced a consistent trend in reallocation of value-added shares towards low-labor-share firms throughout the period, as in the

⁹The details of decomposition are presented in Appendix B.

¹⁰Interestingly, the aggregate labor share decreased in both manufacturing and non-manufacturing but it does not on aggregate. This is because while reallocation across industries within each macro sectors contributed negatively to each macro sectors aggregate labor share, reallocation from manufacturing to non-manufacturing industries contributed positively to the total aggregate labor share.

US. The upward shift of the labor share distribution has also been documented in the US manufacturing industry by Kehrig and Vincent (2018), but is less significant than in France. As emphasized by Kehrig and Vincent (2018), this decomposition groups firms into labor shares quantiles, which allows us to compare two static equilibria. It is conceptually distinct from standard within and cross firm decompositions, because it abstracts from the contributions of firm entry and exit. We focus on long term shifts in the joint distribution of labor and value added shares, not on the role of entry nor on the trajectories of specific firms.

Correlation of rise in concentration and reallocation of labor shares

We now ask whether variations in industry concentration are correlated with labor share variations within industries. To that end we estimate the industry-level relationship between long-term changes in concentration and labor share. We run the following regression:

$$\Delta\lambda_{jt} = \psi_{\lambda} \Delta Conc_{jt} + FE_t + \epsilon_{jt}, \tag{16}$$

where $\Delta Conc_{jt}$ is the 10-year change of sector j concentration level, proxied by the top 1% of top 5% share of sales and FE_t is a set of time fixed-effects that control for year-specific shocks and $\Delta \lambda_{jt}$ is the 10 year change in industry j labor share.

Table 3 presents the results of these regressions. The first two columns show that long term variation of industry concentration are negatively correlated with long term variation of industry labor shares. This relationship is significant and hold for all proxies of concentration. The first two columns of Table D.1 in the Appendix show that this result holds among both manufacturing and non-manufacturing industries. We find that a 10 percentage point rise in concentration is associated with a 0.7 to 1.1 decline in the weighted average labor share of the industry. These results are close to those documented in the US (Autor et al., 2020).

Next we ask whether this result is driven by a correlation between the rise in concentration and the shift in value-added shares from high to low labor share. We consider two components of the 10-year change of the labor share: the cross-quantile contribution to the labor share variation discussed in the previous paragraph, and the evolution of the average labor share of the 5% firms with the lowest labor share within each industry. We use these components as dependent variable in regression (16).

On the one hand, we find that larger increases in concentration are associated with a more negative contribution of value added share reallocation to the aggregate labor share. All coefficients are negative and significant. Table D.1 in the Appendix shows that this result holds among both manufacturing and non-manufacturing industries. On the other hand, we find a positive correlation between change in concentration and change in the average labor share of low labor share firms, defined as firm with labor share in the bottom 5% of their 3-digit industry. These firms are sometimes referred to in the literature as 'hyper-productive' (Kehrig and Vincent, 2018) or 'superstar' firms (Autor et al., 2020). As we will show next, firms with low labor shares also tend to be larger in our sample. Our result suggests that the negative correlation between labor share and concentration is not driven by the fall in the labor share of these 'superstar' firms. Table D.1 in the Appendix reports the results for manufacturing and non-manufacturing industries: the correlation between variations of concentration and variations of the labor share of low-labor-share firms is mostly positive in manufacturing, indicating that the 'superstar' firms in manufacturing today have not only higher market shares but also higher labor shares. Results for non-manufacturing are mixed and vary with the concentration index, but we do not find a negative correlation that is significant at the 5% level.

Labor share and size

To reconcile these facts, we show that there is a negative correlation between labor share and firm size, and that this relationship is monotonic. We run the following regression:

$$\lambda_{it} = F E_{size_{it}} + F E_{j_{it}} + \epsilon_{it},\tag{17}$$

where $FE_{size_{it}}$ is a set of dummies indicating in the size of firm *i* in industry *j* in terms of employment categories, $FE_{j_{it}}$ is a set of interacted fixed effects at the 3-digit industry *j* and year level.

Figure 5 presents the results of this regression, considering labor share in value added and in gross output. Relative to 10-20 employee firms, larger firms tend to report lower labor shares even after controlling for industry and year fixed effects. This decreasing relationship is monotonic, at all levels of employment. Labor shares of firms with 50 to 100 employees tend to be 2 percentage points lower than labor shares of 10 to 20 employees firms of the same industry at the same year. For firms with 2500 to 5000 employees the gap rises to 5 percentage points considering labor share in value added and to 7 percentage points considering labor share in sales.

This decreasing relationship between firm size and labor share helps reconcile our previous results: in industries where concentration increased, the weighted mean labor share decreased and the distribution of labor shares shifted upwards. The reallocation effect due to rising concentration, drives the change in the weighted mean labor share, as larger firms who gain market shares also have lower labor shares.

Firm level trends

So far, we have focused on describing long term shifts in the distribution of firm outcomes, without discussing whether these firms are the same over time. In what follows, we look at the within-firm variation of labor share, for different groups of firm size. We run the following regression:

$$\lambda_{it} = FE_i + \Psi_\lambda t + Control_{it} + \epsilon_{it},\tag{18}$$

where λ_{it} is the labor share of firm *i* in year *t*, FE_i is a firm fixed effect, and $Control_{it}$ is either the logarithm of employment or a set of categories of employment size fixed effects. We run this regression on four samples, with different thresholds of employment size, and two panels, an unbalanced panel of firms possibly entering and exiting, and a balanced panel of firms that are present in the sample from 1984 to 2016.

Table 4 presents the results of these regressions. We find that average firm experienced a trend increase in labor share of around 0.6 percentage points per year, including controlling for changes in the employment level. Firms that are present in the sample from 1984 to 2016 experienced a trend increase of around 0.25 percentage points per year. If we restrict our sample to firms with more than 50, 100, or 1000 employees, we find results that are similar across all specifications and panels: the average firm of any size experienced a trend increase in labor share of around 0.3 percentage points per year. Finally, for larger firms, we find a negative within-firm correlation between the logarithm of employment and the labor share, which complements the negative correlation across firms reported in Figure 5: as a firm's level of employment grows, its labor share decreases.¹¹ This negative relationship is not verified for smaller firms, many of which are sole-proprietorship firms that tend to increase salaried employment relative to non-salaried employment as they grow.

6 Estimation Results

In this section, we first present the results of our estimation procedure, and then show how aggregate and firm-level markups have evolved in France. We document additional facts about market power and concentration, and how variations in market power have contributed to the aggregate labor share, compared to other technological factors.

Production function

Table 5 reports the results of the non-rolling estimation of the production and Table 6 reports the results of the rolling estimation of the production function, for the 27 sectors of our data.¹² Rolling estimations are obtained by first estimating the parameters of the production function $\beta_{jt} \in$ $\{\beta_{l,j}; \beta_{k,j}; \beta_{ll,j}; \beta_{kk,j}; \beta_{lk,j}\}$ in industry j on 11-year rolling window samples, and then averaging for each firm each year the various estimated output elasticities based on samples that include that year:

¹¹Several papers (Lashkari et al., 2019; Autor et al., 2020) have put forward a theoretical explanation of the negative relationship between labor share and firm size, both in the cross-section and within firms.

¹²Estimation is done at the 27 sector level. Each sector includes several 3-digit industries.

$$\beta_{jt}^{\text{rolling}} = \frac{1}{11} \sum_{n=-5}^{5} \beta_j^{t+n},$$

where β_j^t is the estimated parameter on the sample restricted to years t - 5 to t + 5. For the first and last five years of our sample, the average is calculated on fewer estimates. Unlike in a Cobb–Douglas production function, output elasticities also vary across firms of the same sector, even in a given year. We report, for the different sectors, the average and the standard deviation of the elasticities of labor and capital across firms.¹³ Since the returns to scale vary across firms, it is possible for many firms in a sector to have increasing returns to scale, while the estimate of the industry average returns to scale is close to 1. On average, the output elasticity of labor in our data is 0.74 according to the non-rolling estimates and 0.72 according to the rolling estimates.

Aggregate markup

Figure 6 reports the evolutions of the value-added weighted and unweighted average markups across all firms in our sample for both sets of estimates. The unweighted average markup is smaller than the weighted average markup, because firms with larger value-added have on average higher markup. As we will see below, the positive relationship between markup and size holds in our data, with employment as a measure of size.

We find that according to both non-rolling and rolling estimates, the unweighted average markup has decreased. Overall, the unweighted average markup has fallen from 1.3 to 1.0. The valueadded weighted markup has decreased from 1.6 to 1.4 according to non-rolling-window estimates. When we allow the parameters of the translog production function to vary over time, we report an increase of the weighted average markup from 1.4 to 1.6. This suggests sizeable variations in the translog elasticities over time and shows that constraining them to be constant over time is not a valid assumption.

Figure E.4 in the Appendix presents the levels of the unweighted and weighted average markups for firms in the manufacturing and non-manufacturing sectors separately. Patterns are qualitatively similar in both sectors to what they are in the overall economy. We find a decrease in unweighted markups for both non-rolling and rolling estimations both in manufacturing and non-manufacturing sectors, that broadly matches the one observed in the whole economy. Similarly, the variations of the weighted average markup estimates observed in both manufacturing and non-manufacturing, and according to both sets of estimates, quantitatively matches the one observed in the whole economy. These aggregate patterns are therefore neither specific to one sector nor driven by a reallocation

 $^{^{13}}$ Because the translog production function imposes that the heterogeneity of output elasticities across firms within a given sector is entirely explained by the heterogeneity of labor and capital demand, in a log-linear manner, we note that a few sectors appear to have negative average capital elasticities or low returns to scale. These results are similar to those obtained by De Loecker et al. (2016), who estimate negative average capital elasticities from translog production functions in 3 out of 11 manufacturing indutries. Tables D.2 and D.3 in the Appendix report the median output elasticities which are less influenced by outliers.

from manufacturing to non-manufacturing. Interestingly, our results show that, regardless of the pre-crisis trend, the weighted average markup of manufacturing firms sharply dropped after 2008, and only recovered its pre-crisis level at the end of the period.

Figure E.5 in Appendix shows how relying on the proxy-variable method in Ackerberg et al. (2015) (ACF) would have changed our results. The level of the estimated markups differ, because DP estimates of the output elasticity of labor are on average lower than the ACF estimates. Second, in non-rolling estimation, the unweighted markup is not always larger than the weighted markup, suggesting that the increasing relationship between size and markups is not verified. Third, the trend of the average ACF estimated aggregate markup, is significantly different from the average DP estimated aggregate markup. For instance, with rolling estimations, average markup remained however broadly stable around 1.6 according to the ACF estimates. As discussed in Section 2, we do not favor estimates used with this method.

Markup decomposition

Figure 7 shows the decomposition of the aggregate - weighted average - markup into within-markupquantile and across-markup-quantile components. It shows the importance of controlling for industry and disentangling the respective contributions of variations in value-added shares holding markup constant or in markup holding value-added shares constant to interpret aggregate variation.

The decomposition of the aggregate markup mirrors the decomposition of the aggregate labor shares shows how the within-markup quantile component contributed negatively to the evolution of the aggregate markup, while the cross-quantile contributed positively.¹⁴ The contribution of reallocation across industries is negligible. Firms with relatively higher markups within narrowly defined industries have been gaining value-added shares, while the typical firm markup has slightly decreased. Figure E.6 in the Appendix shows that the reallocation from low to high markup firms holds both in manufacturing and non-manufacturing separately.

Comparing rolling and non-rolling window estimations shows why accounting for variations over time of the production function parameters matter. Even though both estimations yield qualitatively similar contributions of the within and across terms, the within term dominates with rolling estimations and the cross term dominates with non-rolling estimations. This quantitative differences translate into qualitative difference in the aggregate trends of markups estimated with both estimations.

Markup and concentration

As for the labor share, we ask whether the observed rise in concentration is correlated with markup variations, on aggregate or along the distribution of markups. To that end we estimate the industrylevel relationship between long term changes in concentration and the industry aggregate markup,

 $^{^{14}\}mathrm{For}$ ACF, this contribution decreased after 2005.

or the contributions to the aggregate variation. We run the following regressions:

$$\Delta \mu_{jt} = \psi_{\mu} \Delta Conc_{jt} + FE_t + \epsilon_{jt}, \tag{19}$$

where $\Delta \mu_{jt}$ is the 10-year change of sector j aggregate markup level, or one of its contributions according to the decomposition described in Appendix B. The independent variables are the same in Equation (16).

Table 7 reports the results of the estimation of Equation (19). The first two columns show that there is a positive and significant long-term relationship between the evolution of aggregate markup and the evolution of concentration at the 3-digit industry level. This relationship is significant and hold for all proxies of concentration. It holds both for markups estimated with rolling and nonrolling estimations. The two first columns of Tables D.4 and D.5 in the Appendix confirm that this positive correlation is separately present for industries belonging to the manufacturing and non-manufacturing sectors.

Next, as for the labor share, we ask whether this result is driven by a correlation between the rise in concentration and the shift in value-added shares from low to high markup firms. The coefficients of the third and fourth columns of Tables 7 are the results of regressions described in Equation (19), where the dependent variable is the cross-quantile component to the evolution of aggregate markup, while in the last two columns the dependent variable is the within-quantile component of firms high markups, defined as firms with the markup in the top 5 % of their 3-digit industry. They show a positive correlation between the rise in concentration and the cross-quantile component of the evolution of the aggregate markup. As for the labor share, this means that the cross-quantile component contributed more to the rise in markup in those industries that have become more concentrated at the top.¹⁵

The fifth and sixth columns of Table 7 find no robust evidence that a rise in concentration is correlated with increases in top markups. The correlations with variations in the top 1% and 5% shares of sales are not significantly positive, the correlations with variations in the shares of the 4 and 20 largest firms are all negative, and significant at the 5% level for three out of four estimations. These results are also observed on the samples of industries belonging to the manufacturing and non-manufacturing sectors separately, except in non-manufacturing, where the correlation of the average top markup with the top 1% and 5% shares is positive and significant, but small relative to the other components. The fact that top markups are not linked with rises in concentration are consistent with theories (see e.g Aghion et al., 2019) where high productivity firms with higher markups, benefit from an external shock more than laggard firms, and expand without increasing their markup. However, it is in contrast with results in the US documented by De Loecker et al. (2020) where top markups contributed to a third of the overall increase in weighted average markups. However, the authors do not provide evidence that the rise in top firm markups is

¹⁵See also Tables D.4 and D.5 of the appendix)

correlated at the industry level with the reallocation component, or with concentration.

Markup and size

As for the labor share, we investigate whether markups are increasing with firm size to understand the correlation between the growing share of the largest firms in each industry's total sales and the reallocation of market shares towards high markup firms. To that end, we run the following regression:

$$\mu_{it} = F E_{size_{it}} + F E_{it} + \epsilon_{it},\tag{20}$$

where $FE_{size_{it}}$ is a set of dummies indicating in the size of firm *i* in industry *j* in terms of employment categories, FE_{jt} is a set of interacted fixed effects at the 3-digit industry *j* and year level.

Figure 8 reports the results of this regression. In our data, we find that larger firms have higher estimated markups. Firms with more than 5000 employees have, on average, markups larger by 30 percentage points than firms with 10 to 20 employees firms, within the same 3-digit industry, on the same year. This increasing relationship is well observed at all levels of employment, and both for markups obtained with the non-rolling and rolling estimations. The top two panels of Figure E.7 in the Appendix show that this result holds both in manufacturing and non-manufacturing sectors separately, although for larger manufacturing firms the relationship flattens out.

The markup is defined in Equation (4) as the ratio of the output elasticity of labor to the labor share. It is important to note that because the output elasticity of labor vary across firms, the markup is not perfectly correlated with the labor share, and therefore the positive relationship between a firm's markup and its size does not flow directly from the negative relationship between its labor share and its size that we document in section 5. The other four panels of Figure E.7 in the Appendix plot the results of the same regression with labor intensity α_{it} and returns to scale γ_{it} as the dependent variable, and shows that the intensity of labor in the production process decreases with size, while returns to scale increase with firm size.

Markup trends

Before turning to the link between the evolution of markups and labor shares in France, we look at the within-firm variation of markups, for different groups of firm size. We run the following regression:

$$\mu_{it} = FE_i + \Psi_{\mu}t + Control_{it} + \epsilon_{it}, \tag{21}$$

where μ_{it} is firm *i* markup in year *t*, FE_i is a firm fixed effect, and $Control_{it}$ is either the logarithm

of employment or a set of categories of employment size fixed effects. We run this regression on four samples, with different thresholds of employment size, and two panels, an unbalanced panel of firms possibly entering and exiting, and a balanced panel of firms that are present in the sample from 1984 to 2016.

Table 8 reports the results for markups obtained with non-rolling estimation and Table 9 for markups obtained with rolling estimation. We find that the average firm experienced a trend decrease in markup of around 1.3 to 1.6 percentage points per year. The decrease in larger firm markups is around 0.8 to 0.9 percentage points per year according to non-rolling estimates, and close to zero according to rolling estimates. For the balanced panel of firms that remain in the sample, markups decreased by 0.6 to 0.9 percentage points per year depending on group of size, according to non-rolling estimates, and close to zero according to rolling estimates, and close to zero according to non-rolling estimates, and close to zero according to rolling estimates. Overall, we find that firm level markups on average decreased, but less so for large and surviving firms. This result indicates that part of the decrease of the within-quantile contribution to aggregate markup is driven by smaller firms and by firm entry and exit.

Link between labor shares and markups

In this section, we circle back to the labor share and ask whether variations in firm-level labor share are mainly driven by markups - i.e. are labor shares increasing because markups are decreasing? - or by technology - i.e. are labor shares increasing because production has become *more* labor intensive?

First, we find that there is a clear negative relationship between firm-level labor shares and markups in France. We run the following regressions:

$$\lambda_{it} = \phi \mu_{it} + F E_{ijt} + \epsilon_{it}, \tag{22}$$

where μ_{it} is the markup of firm *i* in year *t*, λ_{it} is the labor share, and FE_{ijt} is a set of fixed effect, either industry, or firm-level, and year.

Table 10 present the results of these regressions, and show that firms with high markup have low labor shares both across industries and across firms within the same industry. We also find that as a firm markup grows, its labor share decreases. The absolute value of coefficient ϕ is around 0.3 to 0.5 depending on the estimation: as a firm's markup increases 10 percentage points, its labor share decreases by 3 percentage points. Finally, as the coefficient of determination of the regression without fixed effects shows, the heterogeneity of markups explains 45% of the heterogeneity of labor shares across firms. The different panels of the table show that this relationship holds statistically and quantitatively for various groups of size.

To extrapolate these firm-level results to the aggregate economy, we need to keep in mind that there is no such a thing as a representative firm in this context. Recall that Equations (7) and (8) show that at the level of the individual firm, the labor share is the product of labor intensity,

returns to scale and the inverse markup:

$$\lambda_{it} = \alpha_{it} \gamma_{it} \mu_{it}^{-1},$$

but this result does not hold at the aggregate level. From Equation (9), we now decompose variations of the aggregate labor share into contributions from labor intensity, returns to scale, and markups, either by taking the "representative firm" approach and computing the contributions of the weighted averages of each component of the aggregate labor share, therefore ignoring the reallocation between firms; or alternatively, by isolating the contribution of reallocation and computing the contributions of the unweighted averages of each component. Appendix C provides further information on the decomposition.

Figure 9 presents the results of the decomposition for the representative firm. The total variation of the aggregate labor share from 1984 to 2016 is small and positive, and ignoring the role of reallocation, aggregate markups have contributed positively to the aggregate labor share according to the non-rolling estimates, and negatively according to the rolling estimates. This is consistent with evidence above that the conclusions in terms of the variations of the aggregate markup are not the same in both sets of estimates. The sum of the contributions of labor intensity and returns to scale, in other words the contribution of weighted average output elasticity of labor, is negative according to non-rolling estimates and positive according to rolling estimates.

However, taking into account reallocation provides a better picture of underlying determinants of the dynamics of the aggregate labor share in France. Figure 10 presents the results of the decomposition isolating the contribution of reallocation. The contribution of reallocation is negative and very significant, as we have already showed in Figures 4 and 3. In both sets of estimates, firm-level markups have contributed positively to the aggregate labor share, while firm-level returns to scale and labor intensity had a slight negative contribution. Figures E.8 and E.9 show that this results hold in both the manufacturing and non-manufacturing sectors.

7 Conclusion

In this paper, we find no evidence of a rise in market power in France: firm-level markups decreased on average, and the rise in concentration is not correlated with increases in top markups. These facts are however correlated with an important reallocation of market shares towards low-labor share and high-markup firms. Because those firms tend to be larger, this reallocation translates into a rise in concentration.

This reallocation of market shares towards large firms is consistent with a wealth of evidence about the increasing differences between firms (Decker et al., 2016a,c,b; Andrews et al., 2016; Karahan et al., 2019). However, the simultaneous rise in concentration and the relative stability of top firm-level markups raises questions about the interpretation of concentration that goes beyond the French case. One channel that could possibly explain both the reallocation of market shares towards large firms and the within-firm increase in the labor share of income is an increase in winner-take-most competition level, as discussed by Autor et al. (2020): as consumers become more sensitive to firm prices, more productive and bigger firms gain market share but a given firm market power decreases. The source of this increase in competition could be international competition (Bonfiglioli et al., 2019; Panon, 2020), but our results hold across broad sectors of the French economy, including within non manufacturing firms, which suggests that other factors are at play. Technological factors, such as the rise of internet platforms and price comparison websites, may explain why firm-level market power has decreased.

The textbook explanation of a rise in competition has many predictions that are consistent with evidence provided here (Boussard and Lee, 2020). We do not take a stance on the source of market power, and in particular on why there is an increasing relationship between firm size and firm markup: the price elasticity of demand may decrease with quantity, or large firms may be large enough to influence the equilibrium price, and therefore act strategically. However, in both cases, an increase in competition will have offsetting effects on the markup of large firms: holding size constant, it will tend to decrease their markup, but because of reallocation, these firms will grow and their markup will increase. Qualitatively, it is thus possible to observe a rise in top firm markups, as De Loecker et al. (2020) find for the US, or a stability or decrease, as we find for France.

References

- Ackerberg, Daniel A, Kevin Caves, and Garth Frazer, "Identification Properties of Recent Production Function Estimators," *Econometrica*, 2015, 83 (6), 2411–2451. 10, 19
- Aghion, Philippe, Antonin Bergeaud, Timo Boppart, Peter J Klenow, and Huiyu Li, "A Theory of Falling Growth and Rising Rents," *Federal Reserve Bank of San Francisco, Working Paper Series*, 2019, (March), 1–43. 20
- _, Nicholas Bloom, Richard Blundell, Rachel Griffith, and Peter Howitt, "Competition and innovation: an inverted-U relationship," *Quarterly Journal of Economics*, 2005, 120 (2), 701–728.
- Andrews, Dan, Chiara Criscuolo, and Peter Gal, "The Global Productivity Slowdown, Technology Divergence and Public Policy: A Firm Level Perspective," *The Future of Productivity: Main Background Papers*, 2016, pp. 1–50. 4, 13, 23
- Autor, David, David Dorn, Lawrence F. Katz, Christina Patterson, John Van Reenen, and John Van Reenen, "The Fall of the Labor Share and the Rise of Superstar Firms," *The Quarterly Journal of Economics*, 2 2020, 135 (2), 645–709. 3, 5, 12, 13, 15, 16, 17, 24
- Baqaee, David Rezza and Emmanuel Farhi, "Productivity and Misallocation in General Equilibrium," The Quarterly Journal of Economics, 9 2020, 135 (1), 105–163. 3, 8
- Barkai, Simcha, "Declining Labor and Capital Shares," The Journal of Finance, 4 2020, 75 (5), jofi.12909.
 3, 5
- Bauer, Arthur, Bertrand Garbinti, and Simon Georges-Kot, "Financial Constraints and Self-Employment in France, 1945-2014," INSEE, Document de travail de la Dese, 2018, (G2018/08). 3
- Berlingieri, Giuseppe, Patrick Blanchenay, and Chiara Criscuolo, "The great divergence(s)", OECD Science, Technology and Industry," 2017. 4
- Blundell, Richard and Stephen Bond, "GMM Estimation with persistent panel data: an application to production functions," *Econometric Reviews*, 2000, 19 (3), 321–340. 9
- Bonfiglioli, Alessandra, Rosario Crinò, and Gino Gancia, "Concentration in International Markets: Evidence from US Imports," 2019. 3, 24
- Boussard, Jocelyn and Raphaël Lee, "Competition, Profit Share and Concentration," *INSEE*, Document de travail de la Dese, 2020, G2020/04. 3, 5, 24
- Caliendo, Lorenzo, Ferdinando Monte, and Esteban Alejandro Rossi-Hansberg, "The Anatomy of French Production Hierarchies," *Journal of Political Economy*, 2015, 123 (4), 1–75. 11
- Card, David, Jörg Heining, and Patrick Kline, "Workplace Heterogeneity and the Rise of West German Wage Inequality," *Quarterly Journal of Economics*, 2013, 128 (3), 967–1015.
- CEA, "Benefits of competition and indicators of market power," Technical Report, Council of Economic Advisers Issue Brief 2016. 13

- Cette, Gilbert, Lorraine Koehl, and Thomas Philippon, "Labor Shares the Long Term: A Decline?," Economie et Statistique / Economics and Statistics, 2019, (510-511-5), 35-51. 13
- Decker, Ryan A., John C. Haltiwanger, Ron S Jarmin, and Javier Miranda, "Declining Business Dynamism: Implications for Productivity?," 2016. 23
- ___, John Haltiwanger, Ron S. Jarmin, and Javier Miranda, "Declining business dynamism: What we know and the way forward," in "American Economic Review," Vol. 106 2016, pp. 203–207. 23
- _ , _ , _ , **and** _ , "Where has all the skewness gone? The decline in high-growth (young) firms in the U.S.," *European Economic Review*, 2016, *86*, 4–23. 23
- **Deroyon, Julien**, "De nouvelles données issues du profilage des groupes," in "INSEE Références Les Enterprises en France" Insee Références 2015, pp. 39–52. 11
- di Giovanni, Julian, Andrei A. Levchenko, and Isabelle Méjean, "Firms, Destinations, and Aggregate Fluctuations," *Econometrica*, 2014, 82 (4), 1303–1340. 11
- **Dobbelaere, Sabien and Jacques Mairesse**, "Panel data estimates of the production function and product and labor market imperfections," *Journal of Applied Econometrics*, 1 2013, 28 (1), 1–46. 7
- _ and Kozo Kiyota, "Labor market imperfections, markups and productivity in multinationals and exporters," *Labour Economics*, 2018, 53, 198–212. 7
- Doraszelski, Ulrich and Jordi Jaumandreu, "Using Cost Minimization to Estimate Markups," 2019. 7, 10
- Elsby, Michael W. L., Bart Hobijn, and Aysegül Sahin, "The decline of the US labor share," *Brookings* Papers on Economic Activity, 2013, Fall (2), 1–63. 3, 4
- Garbinti, Bertrand, Jonathan Goupille-Lebret, and Thomas Piketty, "Income inequality in France, 1900–2014: Evidence from Distributional National Accounts (DINA)," *Journal of Public Economics*, 6 2018, *162*, 63–77. 3
- Grassi, Basile, "IO in I-O: Competition and Volatility in Input-Output Networks," 2017. 8
- Grossman, Gene, Elhanan Helpman, Ezra Oberfield, and Thomas Sampson, "The Productivity Slowdown and the Declining Labor Share," *Society for Economic Dynamics*, 2018 Meeting Papers, 2018, pp. 1–51. 3, 5, 12
- Gutiérrez, Germán and Thomas Philippon, "Investmentless growth: An empirical investigation," Brookings Papers on Economic Activity, 2017, Fall (1), 89–190. 3
- and _ , "How EU Markets Became Free: A Study of Institutional Drift," National Bureau of Economic Research Working Paper, 2018, 24700. 3, 5
- Hopenhayn, Hugo A, "Entry, Exit, and firm Dynamics in Long Run Equilibrium," Econometrica, 1992, 60 (5), 1127–1150. 5
- Karabarbounis, Loukas and Brent Neiman, "The Global Decline of the Labor Share," Quarterly Journal of Economics, 2014, 129 (1), 61–103. 3, 5, 12

- and _ , "Accounting for Factorless Income," National Bureau of Economic Research Working Paper, 2018, (24404). 5
- Karahan, Fatih, Benjamin Pugsley, and Aysegul Sahin, "Demographic Origins of the Startup Deficit," Federal Reserve Bank of New York Staff Reports, 5 2019, 888. 23
- Kehrig, Matthias and Nicolas Vincent, "The Micro-Level Anatomy of the Labor Share Decline," National Bureau of Economic Research Working Paper, 2018, 25275. 3, 12, 14, 15, 48
- Lashkari, Danial, Arthur Bauer, and Jocelyn Boussard, "Information Technology and Returns to Scale," 2019. 4, 17
- Levinsohn, James, Amil Petrin, and James Levinsohn, "Estimating Production Functions Using Inputs to Control for Unobservables," *Review of Economic Studies*, 2003, 70 (2), 317–341. 10
- Loecker, Jan De and Frederic Warzynski, "Markups and Firm-Level Export Status," American Economic Review, 2012, 102 (6), 2437–2471. 4, 6, 7
- _, Jan Eeckhout, and Gabriel Unger, "The Rise of Market Power and the Macroeconomic Implications," The Quarterly Journal of Economics, 1 2020, 135 (2), 561–644. 3, 20, 24
- _ , Pinelopi K. Goldberg, Amit K. Khandelwal, and Nina Pavcnik, "Prices, Markups, and Trade Reform," *Econometrica*, 2016. 18
- Mayer, Thierry, Marc J. Melitz, and Gianmarco I. P. Ottaviano, "Market size, competition, and the product mix of exporters," *American Economic Review*, 2014, 104 (2), 495–536. 3
- Melitz, Marc J., "The Impact of Trade on Intra-Industry Reallocations and Aggregate Industry Productivity," *Econometrica*, 2003, 71 (6), 1695–1725. 3, 5
- Olley, G. Steven and Ariel Pakes, "The Dynamics of Productivity in the Telecommunications Equipment Industry," *Econometrica*, 1996, 64 (6), 1263. 10
- Panon, Ludovic, "Labor Share, Foreign Demand and Superstar Exporters," 2020. 3, 24
- Piketty, Thomas, Emmanuel Saez, and Gabriel Zucman, "Distributional national accounts: Methods and estimates for the United States," *Quarterly Journal of Economics*, 5 2018, 133 (2), 553–609.
- Reenen, John Van, "Increasing Differences between firms: Market Power and the Macro-Economy," Center for Economic Performance, Discussion Paper 1576, 2018. 3
- Song, Jae, David J Price, Fatih Guvenen, Nicholas Bloom, and Till von Wachter, "Firming Up Inequality," *Quarterly Journal of Economics*, 2018, 134 (1), 1–50. 4
- van Ark, Bart and Kirsten Jäger, "Recent Trends in Europe's Output and Productivity Growth Performance at the Sector Level, 2002-2015," International Productivity Monitor, 2017, 33 (1), 8–23. 11

Tables 8

	Obs	Labor Costs Total Share		Value A	Added	Investment		
				Total	Share	Total	Share	
	(Nb)	$(\mathbf{M} \Subset)$	(%)	$(\mathbf{M} \Subset)$	(%)	$(\mathbf{M} \Subset)$	(%)	
1984	532,996	283,772	82	283,772	84	47,202	70	
1985	$548,\!669$	$312,\!930$	84	$312,\!930$	87	49,752	68	
1986	$571,\!885$	$332,\!184$	84	$332,\!184$	84	$57,\!344$	71	
1987	$592,\!065$	$351,\!970$	84	$351,\!970$	84	56,737	65	
1988	$601,\!927$	379,705	83	379,705	82	$63,\!893$	65	
1989	596,754	$413,\!480$	84	$413,\!480$	83	$73,\!858$	67	
1990	$647,\!678$	$428,\!452$	83	$428,\!452$	81	$78,\!487$	65	
1991	666,606	$458,\!394$	84	$458,\!394$	83	$81,\!559$	64	
1992	702,357	$471,\!285$	85	$471,\!285$	82	81,502	65	
1993	$734,\!122$	$475,\!615$	86	$475,\!615$	84	77,745	67	
1994	$741,\!347$	487,676	86	$487,\!676$	84	$73,\!961$	63	
1995	$765,\!457$	$510,\!294$	87	$510,\!294$	84	79,238	66	
1996	796,722	$515,\!257$	86	$515,\!257$	84	79,844	65	
1997	868,408	$544,\!548$	88	$544,\!548$	85	$112,\!410$	90	
1998	$851,\!193$	$575,\!456$	89	$575,\!456$	85	90,792	67	
1999	$852,\!305$	$607,\!464$	89	$607,\!464$	87	97,016	66	
2000	$913,\!683$	651, 199	89	651, 199	87	$120,\!356$	74	
2001	$891,\!453$	672,645	89	$672,\!645$	86	120,600	70	
2002	$925,\!390$	$696,\!835$	89	$696,\!835$	86	$110,\!329$	65	
2003	$938,\!783$	707,062	89	707,062	85	120,849	71	
2004	976,069	739,259	89	739,259	85	$121,\!434$	68	
2005	$991,\!904$	770,758	89	770,758	86	$146,\!352$	78	
2006	1,040,977	809,623	89	809,623	86	$128,\!399$	63	
2007	1,058,540	845,743	89	845,743	85	169,717	75	
2008	1,022,553	880,096	92	880,096	86	$187,\!424$	78	
2009	$991,\!614$	830,123	90	830,123	84	$148,\!316$	70	
2010	$984,\!428$	864,506	89	864,506	86	$150,\!904$	68	
2011	$947,\!166$	874,459	88	874,459	84	$154,\!229$	65	
2012	$944,\!272$	875,717	88	875,717	83	$158,\!964$	66	
2013	$943,\!845$	882,930	87	882,930	82	$174,\!412$	72	
2014	$937,\!468$	888,054	86	888,054	82	$148,\!430$	60	
2015	$952,\!305$	$911,\!883$	86	911,883	81	$158,\!674$	62	
2016	1,061,582	940,008	87	940,008	82	$146,\!943$	55	
1984 - 2016	$836,\!137$	$636,\!042$	87	636,042	84	$111,\!142$	68	

Table 1: Data Representativeness

Note: This table presents the share of aggregate labor costs (including employer social contributions), value-added and investment in the corporate sector in France that our sample accounts for year by year and on average over the whole period, in total and in percentage of values reported in national accounts. The sample is all firms in the corporate market sectors, excluding agriculture, finance and real estate, with non zero employment.

	Obs. (Nb)	Mean	Median	Sd
Sales	27,543,090	2,642.620	284.620	77,556.280
Gross Output	$27,\!517,\!472$	1,818.489	203.735	$69,\!157.491$
Value-Added	$27,\!517,\!472$	730.007	111.297	$32,\!121.450$
Labor Costs	$27,\!517,\!428$	507.781	81.000	$18,\!092.476$
Labor Share	$27,\!334,\!884$	0.751	0.741	0.336
Employment	$27,\!360,\!292$	14.115	3.000	471.567
Intermediary Inputs	$27,\!517,\!477$	$1,\!088.481$	80.188	$46,\!270.444$
Investment	$19,\!814,\!136$	185.104	4.000	$19,\!200.450$
Capital Book Value	27,507,848	1,305.843	76.000	168,002.986

 Table 2: Summary Statistics

Note: This table presents the main descriptive statistics of firm level outcomes. The sample is all firms in the corporate market sectors, excluding agriculture, finance and real estate, with non zero employment. Values are in thousand euros, except employment which is the average number of full-time equivalent salaried workers.

	Industry Labor Share		Labor	Across Labor Share Quantiles		Within Low Labor Share Quantiles	
Labor Share							
Top 1% Share	-0.0777		-0.0457		0.0097		
	(0.0123)		(0.0112)		(0.0099)		
Top 5% Share		-0.1102		-0.1288		0.0092	
		(0.0167)		(0.0150)		(0.0135)	
Observations	4,666	4,673	4,665	4,660	4,661	4,664	
R2	0.0341	0.0347	0.0290	0.0405	0.0281	0.0292	
Labor Share							
4 Largest Share	-0.0728		-0.0602		0.0772		
	(0.0147)		(0.0133)		(0.0119)		
20 Largest Share		-0.1113		-0.1196		0.0615	
		(0.0168)		(0.0152)		(0.0137)	
Observations	4,649	4,648	4,645	4,645	4,651	4,650	
R2	0.0320	0.0388	0.0325	0.0401	0.0366	0.0340	

Table 3: Correlations Between Variations in Industry-Level Concentration and Labor Shares

Note: Each estimate is the result of OLS estimation at the 3-digit industry with year fixed-effects. The dependent variable in columns "Industry Labor Share" is the long-term change of the industry aggregate labor share, defined as the ratio of the sum of firm level compensation and taxes paid on labor over the sum of firm level value added in that industry. The dependant variable in columns "Across Labor Share Quantiles" and "Within Low Labor Share Quantiles" are the corresponding contributions to the industry aggregate labor share according to the decomposition described in Appendix B, where low quantiles are the bottom 5%. The independent variables are the changes of the share of sales of the top 1%, top 5 %, largest 4 and largest 20 firms.

Labor Share		Unbalanced Panel		Balanced Panel				
	Firm FE	Firm x Size FE	Firm FE	Firm FE	Firm x Size FE	Firm FE		
No Size Threshold								
Trend	0.0057	0.0062	0.0056	0.0025	0.0026	0.0025		
	(0.0000)	(0.0000)	(0.0000)	(0.0000)	(0.0001)	(0.0000)		
Log Employment			0.0322			0.0016		
			(0.0002)			(0.0011)		
Observations	26,761,933	26,032,310	26,623,375	887,205	880,534	887,201		
R2	0.594	0.630	0.598	0.429	0.514	0.429		
More than 50 Employees								
Trend	0.0038	0.0042	0.0041	0.0029	0.0030	0.0030		
	(0.0001)	(0.0001)	(0.0001)	(0.0001)	(0.0001)	(0.0001)		
Log Employment			-0.0226			-0.0175		
			(0.0014)			(0.0036)		
Observations	849,448	803,590	818,020	$165,\!843$	$165,\!013$	165,843		
R2	0.608	0.630	0.577	0.513	0.587	0.514		
More than 100 Employees								
Trend	0.0037	0.0039	0.0040	0.0029	0.0030	0.0030		
	(0.0001)	(0.0001)	(0.0001)	(0.0002)	(0.0002)	(0.0002)		
Log Employment			-0.0265			-0.0220		
			(0.0022)			(0.0052)		
Observations	434,631	399,285	404,322	95,311	94,916	$95,\!311$		
R2	0.649	0.642	0.597	0.538	0.602	0.539		
More than 1000 Employees								
Trend	0.0032	0.0033	0.0033	0.0028	0.0030	0.0029		
	(0.0004)	(0.0005)	(0.0004)	(0.0006)	(0.0006)	(0.0006)		
Log Employment			-0.0321			-0.0312		
			(0.0088)			(0.0184)		
Observations	$56,\!186$	$26,\!560$	26,760	9,406	9,383	9,406		
R2	0.821	0.719	0.689	0.668	0.716	0.669		

Table 4: Firm Level Labor Share Trends

Note: Each estimate is the result of OLS estimation of firm level labor share on time trends, for four samples: all firms, firms with more than 50 employees, 100 employees, and 1000 employees; and two panels: all firms or a balanced panel of firms present in the data from 1984 to 2016 (these firms account of 20 to 25 % of total value-added). All regressions include a set of firm-level fixed effect. Columns "Firm x Size FE" also include a set of size category fixed effect. Standard errors are clustered at the firm level.

	$ heta_l$	$ heta_k$	Ν		θ_l	$ heta_k$	Ν
Mining	0.607	0.297	45,698	Gas and electricity	0.677	0.231	22,243
	(0.048)	(0.081)			(0.193)	(0.169)	
Food products	0.759	0.130	$1,\!277,\!913$	Water supply and waste	0.652	0.183	$118,\!249$
	(0.053)	(0.100)			(0.141)	(0.125)	
Textiles	0.588	0.111	$282,\!598$	Construction	0.649	0.057	$4,\!969,\!117$
	(0.136)	(0.048)			(0.145)	(0.082)	
Wood, paper and printing	0.813	0.041	552,510	Wholesale and retail trade	0.758	0.086	$8,\!502,\!337$
	(0.118)	(0.105)			(0.171)	(0.138)	
Coke and refined petroleum	0.736	0.323	2,472	Transportation	0.830	0.049	$988,\!348$
	(0.250)	(0.074)			(0.151)	(0.145)	
Chemicals	0.819	0.156	$62,\!567$	Accomodation and food services	0.601	0.184	$3,\!076,\!031$
	(0.059)	(0.073)			(0.151)	(0.128)	
Pharmaceuticals	0.901	0.050	$11,\!657$	Publishing and motion pictures	1.033	0.010	309,540
	(0.344)	(0.295)			(0.237)	(0.214)	
Rubber and plastic products	0.774	0.119	$245,\!896$	Telecommunications	1.089	-0.055	$25,\!191$
	(0.150)	(0.164)			(0.187)	(0.213)	
Basic Metals	0.729	0.108	$545,\!742$	ICT	0.938	-0.016	$324,\!622$
	(0.131)	(0.094)			(0.128)	(0.135)	
Computers and electronics	0.764	0.104	$110,\!072$	Legal, accounting and engineering	0.859	-0.025	$1,\!499,\!590$
	(0.071)	(0.023)			(0.150)	(0.144)	
Electrical equipments	0.750	0.135	$50,\!476$	Scientific research	0.935	0.055	30,461
	(0.026)	(0.048)			(0.242)	(0.211)	
Machinery and equipments	0.839	0.073	$161,\!603$	Advertising and market research	0.998	-0.103	406,636
	(0.071)	(0.046)			(0.091)	(0.092)	
Transport equipments	0.836	0.115	$71,\!000$	Administrative and support	0.746	0.044	$1,\!401,\!753$
	(0.159)	(0.139)			(0.120)	(0.157)	
Other manufacturing products	0.797	0.008	$650,\!254$	Total	0.736	0.078	25,744,576
	(0.089)	(0.073)			(0.175)	(0.137)	

 Table 5: Average Output Elasticities, Non Rolling Estimation

Note: This table reports the output elasticities from non rolling estimation of the translog production function. Columns θ_l and θ_k report the average estimated output elasticity with respect to each factor of production for the translog production function for all firms. Column N report the number of observations in each sector. Standard deviations across firms (not standard errors) of the output elasticities are reported in brackets.

	$ heta_l$	$ heta_k$	Ν		θ_l	$ heta_k$	Ν
Mining	0.611	0.289	$45,\!698$	Gas and electricity	0.697	0.236	22,243
	(0.199)	(0.162)			(0.190)	(0.174)	
Food products	0.754	0.127	$1,\!277,\!913$	Water supply and waste	0.630	0.204	$118,\!249$
	(0.052)	(0.104)			(0.178)	(0.146)	
Textiles	0.553	0.135	$282,\!598$	Construction	0.611	0.078	$4,\!969,\!117$
	(0.221)	(0.157)			(0.175)	(0.087)	
Wood, paper and printing	0.794	0.044	$552,\!510$	Wholesale and retail trade	0.762	0.093	$8,\!502,\!337$
	(0.110)	(0.104)			(0.175)	(0.145)	
Coke and refined petroleum	0.533	0.251	2,472	Transportation	0.840	0.045	$988,\!348$
	(0.391)	(0.258)			(0.156)	(0.148)	
Chemicals	0.806	0.163	$62,\!567$	Accomodation and food services	0.592	0.181	$3,\!076,\!031$
	(0.143)	(0.122)			(0.174)	(0.133)	
Pharmaceuticals	0.898	0.072	$11,\!657$	Publishing and motion pictures	1.077	-0.001	$309,\!540$
	(0.359)	(0.286)			(0.245)	(0.215)	
Rubber and plastic products	0.763	0.125	$245,\!896$	Telecommunications	1.048	-0.035	$25,\!191$
	(0.159)	(0.176)			(0.242)	(0.217)	
Basic Metals	0.719	0.111	$545,\!742$	ICT	0.921	0.002	$324,\!622$
	(0.128)	(0.095)			(0.140)	(0.140)	
Computers and electronics	0.747	0.095	$110,\!072$	Legal, accounting and engineering	0.843	-0.020	$1,\!499,\!590$
	(0.084)	(0.068)			(0.164)	(0.150)	
Electrical equipments	0.766	0.127	$50,\!476$	Scientific research	0.856	0.015	30,461
	(0.136)	(0.101)			(0.259)	(0.230)	
Machinery and equipments	0.808	0.094	$161,\!603$	Advertising and market research	0.867	-0.067	406,636
	(0.137)	(0.069)			(0.269)	(0.140)	
Transport equipments	0.834	0.121	$71,\!000$	Administrative and support	0.757	0.039	$1,\!401,\!753$
	(0.180)	(0.156)			(0.126)	(0.165)	
Other manufacturing products	0.745	0.042	$650,\!254$	Total	0.724	0.086	25,744,576
	(0.129)	(0.080)			(0.193)	(0.143)	

Table 6: Average Output Elasticities, Rolling Estimation

Note: This table reports the output elasticities from rolling estimation of the production function. Columns θ_l and θ_k report the average estimated output elasticity with respect to each factor of production for the translog production function for all firms. Column N report the number of observations in each sector. Standard deviations across firms (not standard errors) of the output elasticities are reported in brackets.
		ıstry 'kup	Mai	ross rkup ntiles	High M	thin Aarkup ntiles	
Markup, Non Rolling							
Top 1% Share	0.1754		0.1205		0.0143		
	(0.0157)		(0.0140)		(0.0112)		
Top 5% Share		0.2416		0.2207		0.0219	
		(0.0216)		(0.0190)		(0.0153)	
Observations	4,664	4,664	4,670	4,668	4,665	$4,\!669$	
R2	0.0607	0.0609	0.0269	0.0389	0.0346	0.0355	
Markup, Non Rolling							
4 Largest Share	0.1855		0.1653		-0.0952		
	(0.0190)		(0.0170)		(0.0135)		
20 Largest Share		0.2008		0.2288		-0.0823	
		(0.0220)		(0.0193)		(0.0157)	
Observations	4,649	4,649	4,650	4,650	4,655	4,655	
R2	0.0544	0.0532	0.0328	0.0417	0.0451	0.0413	
Markup, Rolling							
Top 1% Share	0.2640		0.0790		0.0092		
	(0.0257)		(0.0245)		(0.0145)		
Top 5% Share		0.3577		0.1460		0.0400	
		(0.0353)		(0.0337)		(0.0199)	
Observations	4,660	4,660	4,654	4,654	4,663	4,663	
R2	0.0569	0.0586	0.0120	0.0140	0.0168	0.0177	
Markup, Rolling							
4 Largest Share	0.2098		0.0995		-0.0536		
	(0.0321)		(0.0298)		(0.0175)		
20 Largest Share		0.1702		0.1101		-0.0242	
		(0.0372)		(0.0346)		(0.0202)	
Observations	4,647	4,646	4,644 4,644		4,650	4,650	
R2	0.0482	0.0447	0.0108	0.0112	0.0172	0.0173	

Table 7: Correlations Between Variations in Industry-Level Concentration and Markup

Note: Each estimate is the result of OLS estimation at the 3-digit industry with year fixed-effects. The dependent variable in columns "Industry Markup" is the long-term change of the industry aggregate markup. The dependant variable in columns "Across Markup Quantiles" and "Within High Markup Quantiles" are the corresponding contributions to the industry aggregate markup according to the decomposition described in Appendix B, where high quantiles are the top 5%. The independent variables are the changes of the share of sales of the top 1%, top 5 %, largest 4 and largest 20 firms.

Markup		Unbalanced Panel			Balanced Panel	
	Firm FE	Firm x Size FE	Firm FE	Firm FE	Firm x Size FE	Firm FE
No Size Threshold						
Trend	-0.0141	-0.0158	-0.0142	-0.0060	-0.0064	-0.0062
	(0.0000)	(0.0000)	(0.0000)	(0.0001)	(0.0001)	(0.0001)
Log Employment			0.0198			0.0650
			(0.0004)			(0.0025)
Observations	$25,\!092,\!615$	$24,\!535,\!649$	$25,\!092,\!615$	879,223	$872,\!598$	879,223
R2	0.617	0.649	0.617	0.556	0.644	0.561
More than 50 Employees						
Trend	-0.0080	-0.0088	-0.0087	-0.0069	-0.0074	-0.0075
	(0.0001)	(0.0001)	(0.0001)	(0.0003)	(0.0003)	(0.0003)
Log Employment			0.0631			0.0621
			(0.0027)			(0.0066)
Observations	789,696	775,795	789,696	163,698	$162,\!859$	$163,\!698$
R2	0.676	0.733	0.678	0.630	0.711	0.632
More than 100 Employees						
Trend	-0.0083	-0.0089	-0.0089	-0.0074	-0.0078	-0.0078
	(0.0002)	(0.0002)	(0.0002)	(0.0004)	(0.0004)	(0.0004)
Log Employment			0.0617			0.0562
			(0.0041)			(0.0099)
Observations	391,061	386,202	391,061	94,072	93,670	$94,\!072$
R2	0.694	0.743	0.696	0.652	0.725	0.654
More than 1000 Employees						
Trend	-0.0093	-0.0096	-0.0098	-0.0094	-0.0102	-0.0097
	(0.0010)	(0.0011)	(0.0010)	(0.0016)	(0.0016)	(0.0015)
Log Employment			0.0798			0.0866
			(0.0186)			(0.0373)
Observations	26,261	26,072	26,261	9,309	9,286	9,309
R2	0.781	0.815	0.782	0.785	0.829	0.787

Table 8: Firm Level Markup Trends, Non Rolling

Note: Each estimate is the result of OLS estimation of markups from non rolling estimates on time trends, for four samples: all firms, firms with more than 50 employees, 100 employees, and 1000 employees; and two panels: all firms or a balanced panel of firms present in the data from 1984 to 2016 (these firms account of 20 to 25 % of total value-added). All regressions include a set of firm-level fixed effect. Columns "Firm x Size FE" also include a set of size category fixed effect. Standard errors are clustered at the firm level.

Markup		Unbalanced Panel	l		Balanced Panel	
	Firm FE	Firm x Size FE	Firm FE	Firm FE	Firm x Size FE	Firm FE
No Size Threshold						
Trend	-0.0132	-0.0155	-0.0134	-0.0023	-0.0029	-0.0025
	(0.0000)	(0.0000)	(0.0000)	(0.0001)	(0.0001)	(0.0001)
Log Employment			0.0264			0.0641
			(0.0004)			(0.0026)
Observations	25,092,615	$24,\!535,\!649$	25,092,615	879,223	872,598	879,223
R2	0.616	0.650	0.616	0.514	0.613	0.519
More than 50 Employees						
Trend	-0.0010	-0.0020	-0.0018	0.0007	0.0001	0.0002
	(0.0001)	(0.0002)	(0.0002)	(0.0003)	(0.0003)	(0.0003)
Log Employment			0.0645			0.0641
			(0.0029)			(0.0071)
Observations	789,696	775,795	789,696	163,698	162,859	163,698
R2	0.661	0.723	0.663	0.594	0.688	0.597
More than 100 Employees						
Trend	-0.0006	-0.0013	-0.0012	0.0011	0.0006	0.0007
	(0.0002)	(0.0002)	(0.0002)	(0.0004)	(0.0004)	(0.0004)
Log Employment			0.0646			0.0590
			(0.0045)			(0.0110)
Observations	391,061	386,202	391,061	94,072	93,670	94,072
R2	0.675	0.730	0.677	0.615	0.702	0.617
More than 1000 Employees						
Trend	0.0023	0.0021	0.0017	0.0033	0.0028	0.0028
	(0.0011)	(0.0011)	(0.0011)	(0.0017)	(0.0017)	(0.0016)
Log Employment			0.0999			0.1296
			(0.0202)			(0.0422)
Observations	26,261	26,072	26,261	9,309	9,286	9,309
R2	0.745	0.784	0.747	0.744	0.798	0.747

Table 9: Firm Level Markup Trends, Rolling

Note: Each estimate is the result of OLS estimation of markups from rolling estimates on time trends, for four samples: all firms, firms with more than 50 employees, 100 employees, and 1000 employees; and two panels: all firms or a balanced panel of firms present in the data from 1984 to 2016 (these firms account of 20 to 25 % of total value-added). All regressions include a set of firm-level fixed effect. Columns "Firm x Size FE" also include a set of size category fixed effect. Standard errors are clustered at the firm level.

Labor Share		Non Rolling			Rolling	
	No FE	Industry FE	Firm FE	No FE	Industry FE	Firm FE
No Size Threshold						
Markup	-0.3487	-0.3713	-0.3575	-0.3173	-0.3520	-0.3370
	(0.0037)	(0.0024)	(0.0027)	(0.0041)	(0.0022)	(0.0027)
Observations	25,554,561	$25,\!554,\!533$	25,092,587	25,554,561	$25,\!554,\!533$	25,092,587
R2	0.449	0.518	0.772	0.407	0.489	0.761
More than 50 Employees						
Markup	-0.4202	-0.4460	-0.4989	-0.4070	-0.4351	-0.4797
	(0.0039)	(0.0034)	(0.0043)	(0.0054)	(0.0035)	(0.0044)
Observations	808,003	807,805	789,488	808,003	807,805	789,488
R2	0.519	0.602	0.816	0.493	0.582	0.805
More than 100 Employees						
Markup	-0.3991	-0.4268	-0.4754	-0.3842	-0.4163	-0.4554
	(0.0033)	(0.0039)	(0.0051)	(0.0053)	(0.0041)	(0.0053)
Observations	398,301	398,018	390,768	398,301	398,018	390,768
R2	0.513	0.614	0.825	0.483	0.594	0.814
More than 1000 Employees						
Markup	-0.3320	-0.3633	-0.4129	-0.3270	-0.3709	-0.3912
	(0.0079)	(0.0077)	(0.0106)	(0.0050)	(0.0077)	(0.0125)
Observations	26,684	$25,\!305$	24,839	26,684	25,305	24,839
R2	0.502	0.721	0.900	0.471	0.710	0.892

Table 10: Correlation between Labor Share and Markup

Note: Each estimate is the result of OLS estimation of firm level labor share on markups, for four samples: all firms, firms with more than 50 employees, 100 employees, and 1000 employees; and two panels: manufacturing and non manufacturing firms. All columns include year fixed effects. Columns "No FE" include no industry nor firm fixed effect. Columns "Industry FE" include 3-digit industry-level fixed effects. Columns "Firm FE" include firm-level fixed effects. Standard errors are clustered at the 3-digit x year industry level.

9 Figures



Figure 1: Aggregate Labor Share in France, 1984-2016.

Note: This figures reports the ratio of employee compensation, including payroll taxes, to total value-added in the market sectors in France. See Section 4 for details on the different measures.



Figure 2: Cumulative Change in Concentration

Note: This figure reports the cumulative change of concentration in sales across 3-digit industries. Sample is firms in the market sectors, excluding agriculture, finance and real estate. Industry changes in concentration are weighted by the share of each industry in total sales.



Figure 3: Distributions of Labor Shares and Value Added

Note: The connected lines (right axis) reflect the raw cross-firm distribution of labor shares. The vertical bars (left axis) reflect the share of industry value added of firms in each unweighted decile of labor share. These distributions are averaged across 3-digit industries using value added weights in a given year, and averaged across 5 year periods.



Figure 4: Decomposition of Aggregate Labor Share

Note: This figures reports the results of decomposition of the aggregate labor share and markup described in Appendix B. Quantiles of labor share are calculated each year within 3-digit industries.



Figure 5: Labor Share and Size

Note: This figure reports the conditional average labor share by firm size, with 99% confidence interval. Averages are conditional on a set of flexible fixed effects constructed from the interaction of 3-digit industry codes and year.





Note: This figures reports the levels of the weighted and unweighted mean markup based on non-rolling and rolling estimation of a translog value-added production function.



Figure 7: Decomposition of Aggregate Markup

Note: This figures reports the results of decomposition of the aggregate markup described in Appendix B. Quantiles of markup are calculated each year within 3-digit industries.





Note: This figure reports the conditional average markup by firm size, with 99% confidence interval. Averages are conditional on a set of flexible fixed effects constructed from the interaction of 3-digit industry codes and year.



Figure 9: Contributions to the Evolution of the Aggregate Labor Share, Representative Firm

Note: This figure reports the decomposition of the variation of the labor share of the representative firm from 1984 to 2016, based on translog non-rolling and rolling value-added estimation of the production function. See section 6 for detail.





Note: This figure reports the decomposition of the variation of the aggregate labor share from 1984 to 2016, including the reallocation term, based on translog non-rolling and rolling value-added estimation of the production function. See section 6 for detail.

A Data

Industry codes

Industry classification has changed over the 1985-2016 period. From 1985 to 1993 the classification in vigor was the NAP. It changed to NAF in 1993, to NAF rév. 1 in 2003 and finally to NAF rév. 2 in 2008. There is no one-to-one correspondence between these classifications. As a result we make the choice to map each NAP industry code to its most often associated NAF industry code. Similarly we map each NAF industry code to its most often associated NAF rév. 1 industry code, and each NAF rév. 1 code to its most often association NAF rév. 2. As a result we are able to associate to each firm for each year its industry code in the NAF rév. 2 classification.

Variable definitions

Our data provide information on total sales of goods, services and merchandises, as well as variations in inventory and immobilized production. For inputs, they provide the book value of tangible and intangible capital, the wage bill and payroll taxes, and the cost of materials, merchandise, and other intermediary inputs. All data on sales, cost of inventory variations and cost of inputs are recorded separately for merchandise and other inputs. We follow definitions from the National Accounts and define output as the sum of immobilized production, variations in inventory, and sales excluding the cost of merchandise; and we define intermediary inputs use as the sum of material expenditures minus inventory variations, and other external inputs. These definitions mean that gross output includes the net margin on merchandise sold, not gross sales of merchandise. Importantly, our data also includes in intermediary inputs the cost of purchased external services. Except for employment, our micro data is denominated in current prices, and we do not observe firm-level prices of intermediary and capital inputs, nor output prices. We deflate nominal values of gross output, intermediary inputs, and capital stock at the NA38 sectors level using price indexes for investment and outputs from the September 2018 release of the INSEE Annual National Accounts.

Data cleaning

We exclude micro-firms and profiled enterprises from the 2008-2016 data. Very high or negative observations of labor share that stem from very low or negative value-added observations relative to the firm average across years are replaced with the average labor share of the firm across years. Concentration measures are computed using sales on the entire sample of firms, labor share decomposition and all subsequent analysis are conducted on the sample of firms with at least one salaried employee. The parameters of the translog production function are estimated using a smaller sample of firms with sales above $1M \in$, positive value-added, intermediary inputs, and capital. We also exclude from the estimation sample firms with wage, labor productivity, or capital per employee in the top or bottom 0.1%.

B Decomposition

This section details the decomposition method we apply to aggregate labor share and aggregate inverse markups.

Industry level decomposition

Let $k \in \{1, \dots, K\}$ be some industry classification (e.g., 3 digits in micro data), M stands for an aggregate measure (labor share or markup). Also, let S_k and M_k stand respectively for the weight of the industry in total value-added or total sales, and the industry average measure. Define for any variable X:

$$\Delta X_t \equiv X_t - X_{t-1}, \ \overline{X}_t \equiv \frac{1}{2} \left(X_t + X_{t-1} \right),$$
$$\Delta_T X \equiv X_T - X_0,$$

where T is the last period and 0 is the first period. Our first decomposition is:¹⁶

$$\Delta_T M \equiv \underbrace{\sum_{t=1}^{T} \sum_k \overline{S}_{kt} \,\Delta M_{kt}}_{\text{within industry}} + \underbrace{\sum_{t=1}^{T} \sum_k \Delta S_{kt} \,\overline{M}_{kt}}_{\text{across industries}}.$$
(23)

This allows us to distinguish the extent to which the aggregate variation in markup or labor is due to a change of industry shares or a within industry variation, irrespective of the sectoral composition of the economy.

Within Industry Decomposition

Next, we focus on changes in the indusry-level measure Our aim is to decompose the changes at the industry level to the changes in the distribution of firm level markup o labor share and the changes in the markup or labor share for the firms of a given quantile. Let y denote firm quantile. We can write the industry-level measure as

$$M_{kt} \equiv \int_{\underline{y}}^{\overline{y}} S_{kt}(y) \ M_{kt}(y) \ dy, \tag{24}$$

¹⁶This is simply because:

$$\Delta (S_t M_t) = S_t \Delta M_t + \Delta S_t M_t$$
$$\Delta_T (SM) = \sum_{t=1}^T \Delta (S_t M_t).$$

where $S_{kt}(y)$ denotes the share of industry-k value added or sales that is in firms of quantile y at time t and $M_{kt}(y)$ denotes the weighted average outcome (labor share or markup) of firms of quantile y in industry k at time t. We can now decompose¹⁷

$$\Delta M_{kt} = \underbrace{\int_{\underline{y}}^{\overline{y}} \overline{S}_{kt}(y) \ \Delta M_{kt}(y) \ dy}_{\text{Within quantile}} + \underbrace{\int_{\underline{y}}^{\overline{y}} \Delta S_{kt}(y) \ \overline{M}_{kt}(y) \ dy}_{\text{Cross quantile}}.$$
(26)

Top firms are defined as low-labor-share firms or high-markup firms. We further decompose the within quantile component into three components: top firms component (for firms with y in the top/bottom 5%), above/below median component (for firms with y above/below the median but not top firms) and the rest. Let y^* be the threshold of size for being among the top firms, and y' the median size. We can write

$$\int_{\underline{y}}^{\overline{y}} \overline{S}_{kt}(y) \ \Delta M_{kt}(y) \ dy = \underbrace{\int_{\underline{y}}^{\underline{y}'} \overline{S}_{kt}(y) \ \Delta M_{kt}(y) \ dy}_{\text{Within Below Median}} + \underbrace{\int_{\underline{y}'}^{\underline{y}^*} \overline{S}_{kt}(y) \ \Delta M_{kt}(y) \ dy}_{\text{Within Above Median}} + \underbrace{\int_{\underline{y}^*}^{\overline{y}} \overline{S}_{kt}(y) \ \Delta M_{kt}(y) \ dy}_{\text{Within Top Firms}}$$
(27)

We now summarize the within-industry component change in aggregate measure into the following components:

1. The cross quantile component:

$$\sum_{t=1}^{T} \sum_{k} \overline{S}_{kt} \int_{\underline{y}}^{\overline{y}} \Delta S_{kt}(y) \ \overline{M}_{kt}(y) \, dy.$$

2. The within quantile component:

$$\sum_{t=1}^{T}\sum_{k}\overline{S}_{kt}\int_{\underline{y}}^{\overline{y}}\overline{S}_{kt}\left(y\right)\,\Delta M_{kt}\left(y\right)dy,$$

which can then be further decomposed to

$$\Delta M_{kt} \equiv \underbrace{\sum_{i \in \Omega_t^-} \overline{S}_{it} \,\Delta M_{it}}_{\text{within firm}} + \underbrace{\sum_{i \in \Omega_{kt}^-} \Delta S_{it} \,\overline{M}_{it}}_{\text{cross firm}} + \underbrace{\left(\sum_{i \in \Omega_{kt}^+} S_{it} M_{it} - \sum_{i \in \Omega_{kt-1}^-} S_{it-1} M_{it-1}\right)}_{\text{net entry}},\tag{25}$$

where again shares are computed within the industry.

¹⁷As emphasized by Kehrig and Vincent (2018), this decomposition is conceptually distinct from standard within and cross firm decompositions. Let Ω_{kt} be the set of firms active in time t, and $\overline{\Omega}_{kt}$ be the set of firms common between time t and t-1, Ω_{kt}^+ the set of new firms at time t, and Ω_{kt}^- the set of firms exiting between time t and t+1. We can then write:

(a) Within top firms component:

$$\sum_{t=1}^{T} \sum_{k} \overline{S}_{kt} \int_{y^*}^{\overline{y}} \overline{S}_{kt} \left(y\right) \Delta M_{kt} \left(y\right) dy,$$

(b) Within middle firms component:

$$\sum_{t=1}^{T} \sum_{k} \overline{S}_{kt} \int_{y'}^{y^*} \overline{S}_{kt} \left(y\right) \Delta M_{kt} \left(y\right) dy,$$

(c) Within bottom firms component:

$$\sum_{t=1}^{T} \sum_{k} \overline{S}_{kt} \int_{\underline{y}}^{y'} \overline{S}_{kt} \left(y\right) \Delta M_{kt} \left(y\right) dy.$$

We can also write the average variations in those three components across all industries, without weighing them to their industry level contributions :

(a) Average top firms component:

$$\sum_{t=1}^{T} \sum_{k} \overline{S}_{kt} \frac{\int_{y^*}^{\overline{y}} \overline{S}_{kt}\left(y\right) \, \Delta M_{kt}\left(y\right) dy}{\int_{y^*}^{\overline{y}} \overline{S}_{kt}\left(y\right) dy}$$

(b) Average middle firms component:

$$\sum_{t=1}^{T} \sum_{k} \overline{S}_{kt} \frac{\int_{y'}^{y^*} \overline{S}_{kt}(y) \Delta M_{kt}(y) dy}{\int_{y'}^{y^*} \overline{S}_{kt}(y) dy}$$

(c) Average bottom firms component:

$$\sum_{t=1}^{T} \sum_{k} \overline{S}_{kt} \frac{\int_{\underline{y}}^{\underline{y}'} \overline{S}_{kt} \left(y \right) \, \Delta M_{kt} \left(y \right) dy.}{\int_{\underline{y}}^{\underline{y}'} \overline{S}_{kt} \left(y \right) dy}$$

C Labor Share, Markup, and Technology

In a first exercise, we do not isolate the contribution of reallocation to the aggregate labor share and write the weighted average mean for a given variable Z:

$$\mathbb{E}_t^{\mathrm{RF}}[Z] \equiv \sum_i S_{it} Z_{it},\tag{28}$$

where RF stands for "representative firm". In a second exercise, we take into account the contribu-

tion of reallocation and write the unweighted average mean for a given variable Z:

$$\mathbb{E}_t^{\mathrm{WR}}[Z] \equiv \frac{1}{N_t} \sum_i Z_{it},\tag{29}$$

where N_t is the total number of firms and WR stands for "with reallocation". Equation (9) can be rewritten using the definition in Equation (28), which gives a decomposition of the aggregate labor share into the markup, labor intensity and returns to scale of the representative firm:

$$\Lambda_t = \mathbb{E}_t^{\mathrm{RF}}[\alpha \gamma \mu^{-1}] = \mathbb{E}_t^{\mathrm{RF}}[\alpha] \times \mathbb{E}_t^{\mathrm{RF}}[\gamma] \times \mathbb{E}_t^{\mathrm{RF}}[\mu^{-1}] + \mathrm{COV}_t^{\mathrm{RF}},\tag{30}$$

or using the definition in Equation (29), which gives a decomposition of the aggregate labor share into a reallocation term, defined by the gap between weighted and unweighted average labor share, and firm-level unweighted average markups, labor intensity and returns to scale:

$$\Lambda_t = \left(\mathbb{E}_t^{\mathrm{RF}}[\alpha \gamma \mu^{-1}] - \mathbb{E}_t^{\mathrm{WR}}[\alpha \gamma \mu^{-1}] \right) + \mathbb{E}_t^{\mathrm{WR}}[\alpha] \times \mathbb{E}_t^{\mathrm{WR}}[\gamma] \times \mathbb{E}_t^{\mathrm{WR}}[\mu^{-1}] + \mathrm{COV}_t^{\mathrm{WR}}, \qquad (31)$$

where in both cases COV_t^{R} , $\text{R} \in (\text{RF}, \text{WR})$ gathers all of the covariance terms. This term is positive when firms that have high levels of labor intensity also have high returns to scale and low markups. For each $\text{R} \in (\text{RF}, \text{WR})$, this quantity is defined by:

$$\begin{split} \mathrm{COV}_t^\mathrm{R} &= \mathrm{cov}_t^\mathrm{R}(\alpha, \gamma, \mu^{-1}) + \mathbb{E}_t^\mathrm{R}[\alpha] \mathrm{cov}_t^\mathrm{R}(\gamma, \mu^{-1}) \\ &+ \mathbb{E}_t^\mathrm{R}[\gamma] \mathrm{cov}_t^\mathrm{R}(\alpha, \mu^{-1}) + \mathbb{E}_t^\mathrm{R}[\mu^{-1}] \mathrm{cov}_t^{\mathrm{RF}}(\alpha, \gamma), \end{split}$$

where for all set of variables $(X^s)_{s \in S}$:

$$\operatorname{cov}_{t}^{\operatorname{RF}}\left((X^{s})_{s\in S}\right) = \mathbb{E}_{t}^{\operatorname{RF}}\left[\prod_{s\in S} (X_{t}^{s} - \mathbb{E}_{t}^{\operatorname{RF}}[X^{s}])\right].$$

Defining $\overline{X_t}$ and $\Delta X_t = (X_t - X_{t-1})$ as:

$$\overline{X_t} = \frac{1}{2}(X_t + X_{t-1})$$
, $\Delta X_t = (X_t - X_{t-1}),$

we can decompose the variation of the product of expectations in Equations (30) and (31) into contributions of the variation in automation, returns to scale and markups:

$$\Delta \mathbb{E}_{t}^{\mathrm{R}}[\alpha] \times \mathbb{E}_{t}^{\mathrm{R}}[\gamma] \times \mathbb{E}_{t}^{\mathrm{R}}[\mu^{-1}] = \underbrace{\frac{\Delta \mathbb{E}_{t}^{\mathrm{R}}[\alpha]}{3} \left(\overline{\mathbb{E}_{t}^{\mathrm{R}}[\gamma] \times \mathbb{E}_{t}[\mu^{-1}]} + 2\overline{\mathbb{E}_{t}^{\mathrm{R}}[\gamma]} \times \overline{\mathbb{E}_{t}^{\mathrm{R}}[\mu^{-1}]} \right)}_{\text{Contribution of Labor Intensity}}$$

$$+\underbrace{\frac{\Delta \mathbb{E}_{t}^{\mathrm{R}}[\gamma]}{3} \left(\overline{\mathbb{E}_{t}^{\mathrm{R}}[\alpha] \times \mathbb{E}_{t}^{\mathrm{R}}[\mu^{-1}]} + 2\overline{\mathbb{E}_{t}^{\mathrm{R}}[\alpha]} \times \overline{\mathbb{E}_{t}^{\mathrm{R}}[\mu^{-1}]}\right)}_{\text{Contribution of Returns to Scale}} +\underbrace{\frac{\Delta \mathbb{E}_{t}^{\mathrm{R}}[\mu^{-1}]}{3} \left(\overline{\mathbb{E}_{t}^{\mathrm{R}}[\alpha] \times \mathbb{E}_{t}^{\mathrm{R}}[\gamma]} + 2\overline{\mathbb{E}_{t}^{\mathrm{R}}[\alpha]} \times \overline{\mathbb{E}_{t}^{\mathrm{R}}[\gamma]}\right)}_{\text{Contribution of Markups}},$$
(32)

for both $R \in (NR, WR)$. By adding to the decomposition in Equation (32) the variation of the covariance term and of the reallocation term if R = WR, we obtain the decomposition of the variation of the aggregate labor share $\Delta \Lambda_t$.

D Tables

		ıstry Share	Labor	coss Share ntiles	Low Lab	thin oor Share ntiles	
Manufacturing							
Top 1% Share	-0.0694		-0.0793		0.0387		
	(0.0188)		(0.0156)		(0.0139)		
Top 5% Share		-0.1178		-0.1355		0.0663	
		(0.0254)		(0.0214)		(0.0190)	
Observations	2,131	$2,\!143$	2,135	$2,\!143$	2,130	$2,\!142$	
R2	0.0805	0.0828	0.0553	0.0641	0.0622	0.0647	
Manufacturing							
4 Largest Share	-0.0955		-0.0852		0.0424		
	(0.0232)		(0.0196)		(0.0177)		
20 Largest Share		-0.1764		-0.1796		0.0008	
		(0.0285)		(0.0239)		(0.0215)	
Observations	2,121	$2,\!147$	2,122	2,148	2,119	$2,\!144$	
R2	0.0816	0.0879	0.0580	0.0705	0.0618	0.0588	
Non Manufacturing							
Top 1% Share	-0.0943		-0.0458		-0.0014		
	(0.0171)		(0.0161)		(0.0139)		
Top 5% Share		-0.1054		-0.1448		-0.0310	
		(0.0238)		(0.0223)		(0.0189)	
Observations	2,202	2,201	2,202	$2,\!196$	2,192	$2,\!190$	
R2	0.0234	0.0192	0.0298	0.0419	0.0354	0.0422	
Non Manufacturing							
4 Largest Share	-0.0601		-0.0629		0.0690		
	(0.0194)		(0.0181)		(0.0156)		
20 Largest Share		-0.1102		-0.1196		0.0597	
		(0.0208)		(0.0198)		(0.0171)	
Observations	2,172	2,145	2,176	2,150	2,163	2,137	
R2	0.0146	0.0272	0.0355	0.0454	0.0448	0.0455	

Table D.1: Correlations Between Variations in Industry-Level Concentration and Labor Shares

Note: Each estimate is the result of OLS estimation at the 3-digit industry with year fixed-effects. The dependent variable in columns "Industry Labor Share" is the long-term change of the industry aggregate labor share, defined as the ratio of the sum of firm level compensation and taxes paid on labor over the sum of firm level value added in that industry. The dependant variable in columns "Across Labor Share Quantiles" and "Within Low Labor Share Quantiles" are the corresponding contributions to the industry aggregate labor share according to the decomposition described in Appendix B, where low quantiles are the bottom 5%. The independent variables are the changes of the share of sales of the top 1%, top 5 %, largest 4 and largest 20 firms.

	$ heta_l$	$ heta_k$	Ν		θ_l	$ heta_k$	Ν
Mining	0.606	0.308	$45,\!698$	Gas and electricity	0.648	0.276	22,243
	(0.048)	(0.081)			(0.193)	(0.169)	
Food products	0.754	0.138	$1,\!277,\!913$	Water supply and waste	0.665	0.182	$118,\!249$
	(0.053)	(0.100)			(0.141)	(0.125)	
Textiles	0.569	0.110	$282,\!598$	Construction	0.643	0.063	$4,\!969,\!117$
	(0.136)	(0.048)			(0.145)	(0.082)	
Wood, paper and printing	0.813	0.048	$552,\!510$	Wholesale and retail trade	0.752	0.098	$8,\!502,\!337$
	(0.118)	(0.105)			(0.171)	(0.138)	
Coke and refined petroleum	0.720	0.341	2,472	Transportation	0.822	0.063	$988,\!348$
	(0.250)	(0.074)			(0.151)	(0.145)	
Chemicals	0.821	0.156	$62,\!567$	Accomodation and food services	0.590	0.197	$3,\!076,\!031$
	(0.059)	(0.073)			(0.151)	(0.128)	
Pharmaceuticals	1.013	0.067	$11,\!657$	Publishing and motion pictures	1.118	-0.005	309,540
	(0.344)	(0.295)			(0.237)	(0.214)	
Rubber and plastic products	0.772	0.127	$245,\!896$	Telecommunications	1.160	-0.083	$25,\!191$
	(0.150)	(0.164)			(0.187)	(0.213)	
Basic Metals	0.734	0.115	545,742	ICT	0.937	-0.015	$324,\!622$
	(0.131)	(0.094)			(0.128)	(0.135)	
Computers and electronics	0.757	0.104	$110,\!072$	Legal, accounting and engineering	0.857	-0.017	$1,\!499,\!590$
	(0.071)	(0.023)			(0.150)	(0.144)	
Electrical equipments	0.749	0.130	$50,\!476$	Scientific research	0.956	0.067	30,461
	(0.026)	(0.048)			(0.242)	(0.211)	
Machinery and equipments	0.842	0.076	$161,\!603$	Advertising and market research	1.006	-0.109	406,636
	(0.071)	(0.046)			(0.091)	(0.092)	
Transport equipments	0.840	0.122	$71,\!000$	Administrative and support	0.737	0.052	$1,\!401,\!753$
	(0.159)	(0.139)			(0.120)	(0.157)	
Other manufacturing products	0.795	0.015	$650,\!254$	Total	0.734	0.086	25,744,576
	(0.089)	(0.073)			(0.175)	(0.137)	

Table D.2: Median Output Elasticities, Non Rolling Estimation

Note: This table reports the output elasticities from non rolling estimation of the translog production function. Columns θ_l and θ_k report the median estimated output elasticity with respect to each factor of production for the translog production function for all firms. Column N report the number of observations in each sector. Standard deviations (not standard errors) of the output elasticities are reported in brackets.

	θ_l	$ heta_k$	Ν		θ_l	$ heta_k$	Ν
Mining	0.612	0.334	$45,\!698$	Gas and electricity	0.688	0.265	22,243
	(0.199)	(0.162)			(0.190)	(0.174)	
Food products	0.750	0.134	$1,\!277,\!913$	Water supply and waste	0.639	0.211	$118,\!249$
	(0.052)	(0.104)			(0.178)	(0.146)	
Textiles	0.540	0.118	$282,\!598$	Construction	0.605	0.084	$4,\!969,\!117$
	(0.221)	(0.157)			(0.175)	(0.087)	
Wood, paper and printing	0.803	0.050	$552,\!510$	Wholesale and retail trade	0.762	0.105	$8,\!502,\!337$
	(0.110)	(0.104)			(0.175)	(0.145)	
Coke and refined petroleum	0.258	0.433	2,472	Transportation	0.835	0.060	$988,\!348$
	(0.391)	(0.258)			(0.156)	(0.148)	
Chemicals	0.808	0.179	$62,\!567$	Accomodation and food services	0.585	0.192	$3,\!076,\!031$
	(0.143)	(0.122)			(0.174)	(0.133)	
Pharmaceuticals	0.981	0.106	$11,\!657$	Publishing and motion pictures	1.163	-0.023	$309,\!540$
	(0.359)	(0.286)			(0.245)	(0.215)	
Rubber and plastic products	0.759	0.136	$245,\!896$	Telecommunications	1.111	-0.056	$25,\!191$
	(0.159)	(0.176)			(0.242)	(0.217)	
Basic Metals	0.726	0.117	545,742	ICT	0.921	0.008	$324,\!622$
	(0.128)	(0.095)			(0.140)	(0.140)	
Computers and electronics	0.756	0.091	$110,\!072$	Legal, accounting and engineering	0.843	-0.012	$1,\!499,\!590$
	(0.084)	(0.068)			(0.164)	(0.150)	
Electrical equipments	0.774	0.142	$50,\!476$	Scientific research	0.881	0.017	30,461
	(0.136)	(0.101)			(0.259)	(0.230)	
Machinery and equipments	0.823	0.093	$161,\!603$	Advertising and market research	0.887	-0.056	$406,\!636$
	(0.137)	(0.069)			(0.269)	(0.140)	
Transport equipments	0.837	0.131	$71,\!000$	Administrative and support	0.753	0.047	$1,\!401,\!753$
	(0.180)	(0.156)			(0.126)	(0.165)	
Other manufacturing products	0.748	0.051	$650,\!254$	Total	0.733	0.096	25,744,576
	(0.129)	(0.080)			(0.193)	(0.143)	

Table D.3: Median Output Elasticities, Rolling Estimation

Note: This table reports the output elasticities from rolling estimation of the production function. Columns θ_l and θ_k report the median estimated output elasticity with respect to each factor of production for the translog production function for all firms. Column N report the number of observations in each sector.Standard deviations (not standard errors) of the output elasticities are reported in brackets.

		ıstry kup	Mai	ross rkup ntiles	High N	thin Aarkup ntiles	
Manufacturing							
Top 1% Share	0.1759		0.1480		-0.0111		
	(0.0242)		(0.0202)		(0.0174)		
Top 5% Share		0.3109		0.2818		-0.0177	
		(0.0329)		(0.0274)		(0.0237)	
Observations	2,122	2,131	2,129	$2,\!138$	$2,\!125$	$2,\!136$	
R2	0.0893	0.0985	0.0543	0.0768	0.0536	0.0573	
Manufacturing							
4 Largest Share	0.2511		0.2081		-0.0788		
	(0.0303)		(0.0258)		(0.0219)		
20 Largest Share		0.3305		0.3267		-0.0433	
		(0.0371)		(0.0314)		(0.0268)	
Observations	2,112	2,138	2,118	2,144	2,113	2,139	
R2	0.0918	0.0944	0.0628	0.0787	0.0580	0.0546	
Non Manufacturing							
Top 1% Share	0.2438		0.1465		0.0534		
	(0.0207)		(0.0201)		(0.0152)		
Top 5% Share		0.2679		0.2296		0.0782	
		(0.0294)		(0.0281)		(0.0210)	
Observations	2,218	2,213	2,211	2,205	2,215	2,212	
R2	0.0809	0.0628	0.0311	0.0367	0.0490	0.0499	
Non Manufacturing							
4 Largest Share	0.1671		0.1565		-0.0541		
	(0.0241)		(0.0230)		(0.0175)		
20 Largest Share		0.1891		0.2202		-0.0643	
		(0.0267)		(0.0251)		(0.0194)	
Observations	2,191 2,16		2,185	$2,\!159$	2,191	2,165	
R2	0.0443	0.0482	0.0286	0.0424	0.0495	0.0507	

Table D.4: Correlations Between Variations in Industry-Level Concentration and Markup, Non Rolling

Note: Each estimate is the result of OLS estimation at the 3-digit industry with year fixed-effects. The dependent variable in columns "Industry Markup" is the long-term change of the industry aggregate markup. The dependant variable in columns "Across Markup Quantiles" and "Within High Markup Quantiles" are the corresponding contributions to the industry aggregate markup according to the decomposition described in Appendix B, where high quantiles are the top 5%. The independent variables are the changes of the share of sales of the top 1%, top 5 %, largest 4 and largest 20 firms.

		ıstry 'kup	Mai	ross rkup ntiles	Within High Markup Quantiles			
Manufacturing Top 1% Share	0.2142 (0.0340)		0.1339		-0.0136			
Top 5% Share	(0.0340)	0.3852 (0.0471)	(0.0299)	0.2882 (0.0417)	(0.0217)	0.0279 (0.0297)		
Observations R2	2,135 0.0959	$2,146 \\ 0.1068$	2,135 0.0697	2,145 0.0755	2,131 0.0704	2,142 0.0675		
Manufacturing 4 Largest Share	0.2977 (0.0435)		0.1667 (0.0390)		-0.0237 (0.0274)			
20 Largest Share	(0.0100)	0.4232 (0.0531)	(0.0000)	0.2449 (0.0476)	(0.0211)	0.0518 (0.0334)		
Observations R2	$2,126 \\ 0.1004$	2,152 0.1067	2,126 0.0636	2,152 0.0672	2,117 0.0632	2,143 0.0665		
Non Manufacturing Top 1% Share	0.3458		0.0906		0.0804			
Top 5% Share	(0.0403)	0.3685 (0.0570)	(0.0381)	0.0591 (0.0535)	(0.0192)	0.0729 (0.0270)		
Observations R2	$2,199 \\ 0.0515$	2,194 0.0393	2,183 0.0406	2,178 0.0370	2,198 0.0267	2,193 0.0216		
Non Manufacturing 4 Largest Share	0.1344 (0.0475)		0.1047 (0.0432)		-0.0756 (0.0223)			
20 Largest Share	(0.0410)	0.0327 (0.0533)	(0.0492)	0.0968 (0.0484)	(0.0220)	-0.1055 (0.0248)		
Observations R2	2,174 0.0248	2,147 0.0222	2,156 0.0422	2,130 0.0411	2,175 0.0259	2,149 0.0278		

Table D.5: Correlations Between Variations in Industry-Level Concentration and Markup, Rolling

Note: Each estimate is the result of OLS estimation at the 3-digit industry with year fixed-effects. The dependent variable in columns "Industry Markup" is the long-term change of the industry aggregate markup. The dependant variable in columns "Across Markup Quantiles" and "Within High Markup Quantiles" are the corresponding contributions to the industry aggregate markup according to the decomposition described in Appendix B, where high quantiles are the top 5%. The independent variables are the changes of the share of sales of the top 1%, top 5 %, largest 4 and largest 20 firms.

Labor Share		Manufacturing		Ν	on Manufacturi	ng	
	No FE	Industry FE	Firm FE	No FE	Industry FE	Firm FE	
No Size Threshold							
Markup	-0.3766	-0.3768	-0.3614	-0.3435	-0.3703	-0.3567	
	(0.0099)	(0.0066)	(0.0087)	(0.0031)	(0.0025)	(0.0028)	
Observations	4,189,494	4,189,478	4,133,837	21,365,067	$21,\!365,\!055$	20,958,750	
R2	0.568	0.590	0.774	0.429	0.506	0.772	
More than 50 Employees							
Markup	-0.4789	-0.5017	-0.5483	-0.4019	-0.4180	-0.4678	
	(0.0045)	(0.0051)	(0.0060)	(0.0041)	(0.0039)	(0.0054)	
Observations	296,014	295,922	291,562	511,989	511,883	497,926	
R2	0.566	0.640	0.806	0.523	0.578	0.822	
More than 100 Employees							
Markup	-0.4490	-0.4765	-0.5204	-0.3823	-0.3988	-0.4420	
	(0.0040)	(0.0058)	(0.0071)	(0.0037)	(0.0047)	(0.0066)	
Observations	157,602	157,514	155,793	240,699	240,504	234,975	
R2	0.551	0.644	0.809	0.527	0.587	0.833	
More than 1000 Employees							
Markup	-0.3782	-0.4376	-0.4396	-0.3241	-0.3387	-0.3949	
	(0.0125)	(0.0148)	(0.0168)	(0.0122)	(0.0084)	(0.0135)	
Observations	10,154	9,347	9,238	16,530	15,958	15,601	
R2	0.504	0.739	0.871	0.556	0.706	0.911	

Table D.6: Correlation between Labor Share and Markup, Non Rolling

Note: Each estimate is the result of OLS estimation of firm level labor share on markups, for four samples: all firms, firms with more than 50 employees, 100 employees, and 1000 employees; and two panels: manufacturing and non manufacturing firms. All columns include year fixed effects. Columns "No FE" include no industry nor firm fixed effect. Columns "Industry FE" include 3-digit industry-level fixed effects. Columns "Firm FE" include firm-level fixed effects. Standard errors are clustered at the 3-digit x year industry level.

Labor Share		Manufacturing		Ν	on Manufacturi	ng
	No FE	Industry FE	Firm FE	No FE	Industry FE	Firm FE
No Size Threshold						
Markup	-0.3625	-0.3719	-0.3566	-0.3100	-0.3485	-0.3338
	(0.0090)	(0.0061)	(0.0081)	(0.0035)	(0.0024)	(0.0029)
Observations	4,189,494	4,189,478	4,133,837	21,365,067	$21,\!365,\!055$	20,958,750
R2	0.532	0.567	0.767	0.387	0.475	0.760
More than 50 Employees						
Markup	-0.4354	-0.4727	-0.5062	-0.4013	-0.4153	-0.4617
	(0.0102)	(0.0061)	(0.0071)	(0.0038)	(0.0041)	(0.0054)
Observations	296,014	295,922	291,562	511,989	511,883	497,926
R2	0.506	0.600	0.783	0.511	0.567	0.817
More than 100 Employees						
Markup	-0.4004	-0.4463	-0.4754	-0.3818	-0.3981	-0.4390
	(0.0074)	(0.0067)	(0.0083)	(0.0047)	(0.0050)	(0.0067)
Observations	157,602	157,514	155,793	240,699	240,504	234,975
R2	0.486	0.604	0.786	0.511	0.577	0.829
More than 1000 Employees						
Markup	-0.3053	-0.3912	-0.3805	-0.3439	-0.3624	-0.4000
	(0.0138)	(0.0188)	(0.0214)	(0.0065)	(0.0077)	(0.0141)
Observations	10,154	9,347	9,238	16,530	$15,\!958$	15,601
R2	0.409	0.691	0.852	0.543	0.707	0.909

Table D.7: Correlation between Labor Share and Markup, Rolling

Note: Each estimate is the result of OLS estimation of firm level labor share on markups, for four samples: all firms, firms with more than 50 employees, 100 employees, and 1000 employees; and two panels: manufacturing and non manufacturing firms. All columns include year fixed effects. Columns "No FE" include no industry nor firm fixed effect. Columns "Industry FE" include 3-digit industry-level fixed effects. Columns "Firm FE" include firm-level fixed effects. Standard errors are clustered at the 3-digit x year industry level.



Figure E.1: Cumulative Change in Concentration

Note: This figure reports the cumulative change of concentration in sales across each 3-digit industry. Sample is firms in the market sectors, excluding agriculture, finance and real estate. Industry changes in concentration are weighted by the share of each industry in total sales the previous year.



Figure E.2: Distributions of Labor Shares and Value Added

Note: The connected lines (right axis) reflect the raw cross-firm distribution of labor shares. The vertical bars (left axis) reflect the share of industry value added of firms in each unweighted decide of labor share. These distributions are averaged across 3-digit industries using value added weights in a given year, and then average across 5 year periods.



Figure E.3: Decomposition of the Cumulative Change in Aggregate Labor Share

Note: This figures reports the results of decomposition of the aggregate labor share described in Appendix B. Quantiles of labor share are calculated each year within 3-digit industries.

Figure E.4: Aggregate Markup



Note: This figures reports the levels of the weighted and unweighted mean markup based on non-rolling and rolling estimation of a translog value-added production function.



Figure E.5: Aggregate Markup - ACF

Note: This figures reports the levels of the weighted and unweighted mean markup based on non-rolling and rolling estimation of a translog value-added production function following the ACF procedure.



Note: This figures reports the results of decomposition of the aggregate markup described in Appendix B. Quantiles of markup are calculated each year within 3-digit industries.



Figure E.7: Correlations with Size

Note: This figure reports the conditional average markup, labor intensity and returns to scale by firm size, with 99% confidence interval. Averages are conditional on a set of flexible fixed effects constructed from the interaction of 3-digit industry codes and year.



Figure E.8: Contributions to the Evolution of the Aggregate Labor Share, Representative Firm

(a) Manufacturing, Non Rolling

(b) Manufacturing, Rolling

Note: This figure reports the decomposition of the variation of the aggregate labor share of the representative firm from 1984 to 2016, based on translog non-rolling and rolling value-added estimation of the production function. See section 6 for detail.





(a) Manufacturing, Non Rolling

(b) Manufacturing, Rolling

Note: This figure reports the decomposition of the variation of the aggregate labor share from 1984 to 2016, including the reallocation term, based on translog non-rolling and rolling value-added estimation of the production function. See section 6 for detail.

Liste des documents de travail de la Direction des Études et Synthèses Économiques

 J. OLIVEIRA-MARTINS, J. TOUJAS-BERNATE Macro-econnic import functions with imperfect connention - An abulication to the F.C. Trade 	 STAPIC STAPIC Les échanges internationaux de services de la France dans le cadre des négociations multila- 	térales du GATT Juin 1992 (1ère version) Novembre 1992 (version finale)	P. SEVESTRE L'économétrie sur données individuelles-	temporelles. Une note introductive H. ERKEL-ROUSSE Le commerce extérieur et l'environnement in-	tréestimation 1992)	N. GREENAN et D. GUELLEC Coordination within the firm and endogenous growth	A. MAGNIER et J. TOUJAS-BERNATE Technology and trade: empirical evidences for the major five industrialized countries	B. CREPON, E. DUGUET, D. ENCAOUA et P. MOHNEN Cooperative, non cooperative R & D and optimal patent life	B. CREPON et E. DUGUET Research and development, competition and innovation: an application of pseudo maximum likelihood methods to Poisson models with heterocheity.	J. TOUJAS-BERNATE J. TOUJAS-BERNATE Commerce international et concurrence impar-	Tatte : developpements recents et implications pour la politique commerciale	on. cooco Durées de chômage et comportements d'offre de travail : une revue de la littérature	 H. ERKEL-ROUSSE Union économique et monétaire : le débat économique 	N. GREENAN - D. GUELLEC / G. BROUSSAUDIER - L. MIOTTI Innovation organisationalle, dynamisme tech-	P. JAILLARD P. JAILLARD Le traité de Maastricht : présentation juridique et historique	J.L. BRILLET Micro-DMS : présentation et propriétés J.L. BRILLET	Micro-DMS - variantes : les tableaux S. JACOBZONE Les grands réseaux publics français dans une perspective européenne	L. BLOCH - B. CŒURÉ Profitabilité de l'investissement productif et transmission des chocs financiers
G 9202	G 9203		G 9204	G 9205		G 9206	G 9207	G 9208	G 9209	G 9301	G 0300	2006 0	G 9303	G 9304	G 9305	G 9306 G 9307	G 9308	G 9309
 J. FAYOLLE et M. FLEURBAEY Accumulation, profitabilité et endettement des entreprises 	H. ROUSSE Détection et effets de la multicolinéarité dans les modèles linéaires ordinaires - Un prolongement de la réflexion de BELSLEY, KUH et WELSCH	P. RALLE et J. TOUJAS-BERNATE Indexation des salaires : la rupture de 1983	D. GUELLEC et P. RALLE Compétitivité, croissance et innovation de produit	P. RALLE et J. TOUJAS-BERNATE Les conséquences de la désindexation. Analyse dans une maquette prix-salaires	Équipe AMADEUS Le modèle AMADEUS - Première partie - Présentation rénérale	J.L. BRILLET J.L. BRILLET Leonoide AMADEUS - Deuxième partie - Promidiás variantielles	D. GUELLEC et P. RALLE Endogenous growth and product innovation	H. ROUSSE Le modèle AMADEUS - Troisième partie - Le commerce extérieur et l'environnement international	H. ROUSSE Effets de demande et d'offre dans les résultats du commerce extérieur manufacturé de la France au cours des deux demières décennies	 B. CREPON Innovation, taille et concentration : causalités et dynamiques 	B. AMABLE et D. GUELLEC Un panorama des théories de la croissance	enaogene M. GLAUDE et M. MOUTARDIER Une évaluation du coût direct de l'enfant de 1979	à 1989 P. RALLE et alii	rtance - Allemagne : performances economiques comparées J.L. BRILLET Micro-DMS NOV PARU	R érateur et mu) : quelques re	 B. CREPON et G. DUREAU Investissement en recherche-développement : analyse de causalités dans un modèle d'accélé- rateur généralisé 	 J.L. BRILLET, H. ERKEL-ROUSSE, J. TOUJAS- BERNATE "France-Allemagne Couplées" - Deux économies vues par une maquette macro-économétrique 	W.J. ADAMS, B. CREPON, D. ENCAOUA Choix technologiques et stratégies de dissuasion d'entrée
G 9001	G 9002	G 9003	G 9004	G 9005	G 9101	G 9102	G 9103	G 9104	G 9105	G 9106	G 9107	G 9108	G 9109	G 9110	G 9111	G 9112	G 9113	G 9201

	Suivi-	analysian -	D. B. CO Invest	de l'in B. DO L'éval	I. KAE	When Urreve	L. BL	D. GC Les tr qualifi	N. GF Techr	lificati	D. GO Persis laires:	D. G Derei	a ree a ree data S. JA	Les li persp NON Prévis	G. CE Le pa Const	Bang Prévis Struct	écono Rappi	J. BO L'octr d'info	A. TO Les ir	P. GE Comp d'alcc globa	2
н	G 9410	G 9411	G 94 IZ	G 9413	G 9414		10 800Z	G 9503	G 9504		G 9505	G 9505 Bis	G 9506		G 9507	G 9601	G 9602	G 9603	G 9604	G 9605	
ī	 J. BOURDIEU - B. COLIN-SEDILLOT Les théories sur la structure optimale du capital : quelques points de repère 	 J. BOURDIEU - B. COLIN-SEDILLOT Les décisions de financement des entreprises françaises : une évaluation empirique des théo- ries de la structure optimale du capital 	L. BLOCH - B. CŒURÉ Q de Tobin marginal et transmission des chocs financiers	Équipes Amadeus (INSEE), Banque de France, Métric (DP) Présentation des propriétés des principaux mo- dèles macroéconomiques du Service Public	B. CREPON - E. DUGUET Research & Development, competition and innovation	 B. DORMONT Quelle est l'influence du coût du travail sur l'emploi ? 	D. BLANCHET - C. BROUSSE Deux études sur l'âge de la retraite	 D. BLANCHET Répartition du travail dans une population hété- rogène : deux notes 	D. EYSSARTIER - N. PONTY AMADEUS - an annual macro-economic model for the medium and long term	G. CETTE - Ph. CUNÉO - D. EYSSARTIER - J. GAUTIÉ	Les effets sur l'emploi d'un abaissement du coût du travail des jeunes	D. DLANVOTET Les structures par âge importent-elles ?	J. GAUTIÉ Le chômage des jeunes en France : problème de formation ou phénomène de file d'attente ? Quelques éléments du débat	P. QUIRION Les déchets en France : éléments statistiques et économiques	D. LADIRAY - M. GRUN-REHOMME Lissage par moyennes mobiles - Le problème des extrémités de série	V. MAILLARD Théorie et pratique de la correction des effets de jours ouvrables	F. ROSENWALD La décision d'investir	 S. JACOBZONE Les apports de l'économie industrielle pour définir la stratégie économique de l'hôpital public 	L. BLOCH, J. BOURDIEU, B. COLIN-SEDILLOT, G. LONGUEVILLE Du défaut de paiement au dépôt de bilan : les	banquiers face aux PME en difficulté D. EYSSARTIER, P. MAIRE Impacts macro-économiques de mesures d'aide au logement - quelques éléments d'évaluation	
	G 9310	G 9311	G 9312	G 9313	G 9314	G 9315	G 9316	G 9317	G 9318	G 9319		و 10 e	G 9402	G 9403	G 9404	G 9405	G 9406	G 9407	G 9408	G 9409	

		:=	
G 9606	C. DOZ - F. LENGLART Factor analysis and unobserved component models: an application to the study of French business surveys	G 9712	E. DUBOIS High Real Interest Saving Investmen sufficient Credibilit
G 9607	N. GREENAN - D. GUELLEC La théorie coopérative de la firme	G 9713	Bilan des activités et Synthèses Écor
G 9608	N. GREENAN - D. GUELLEC Technological innovation and employment reallocation	G 9714	F. LEQUILLER Does the French (state Inflation?
G 9609	Ph. COUR - F. RUPPRECHT L'intégration asymétrique au sein du continent américain : un essai de modélisation	G 9715	X. BONNET Peut-on mettre en baisse des salaire
G 9610	S. DUCHENE - G. FORGEOT - A. JACQUOT Analyse des évolutions récentes de la producti- vité apparente du travail	G 9716	Une étude sur que N. IUNG - F. RUPI Productivité de la
G 9611	X. BONNET - S. MAHFOUZ The influence of different specifications of wages- prices spirals on the measure of the NAIRU: the case of France	G 9717	d'échelle dans le s français E. DUGUET - I. K. Appropriation strai
G 9612	PH. COUR - E. DUBOIS, S. MAHFOUZ, J. PISANI-FERRY The cost of fiscal retrenchment revisited: how strong is the evidence?	G 9718	the patent system analysis at the firm L.P. PELÉ - P. RA Âge de la retraite
G 9613	 A. JACQUOT Les flexions des taux d'activité sont-elles seule- ment conjoncturelles ? 	G 9719	gime général ZHANG Yingxiang Lexique macroécc
G 9614	ZHANG Yingxiang - SONG Xueqing Lexique macroéconomique Français-Chinois	G 9720	chinois-français M. HOUDEBINE -
G 9701	 J.L. SCHNEIDER La taxe professionnelle : éléments de cadrage économique 	G 9721	Mesurer l'influence sation des entrepr A. MOUROUGANI
G 9702	 J.L. SCHNEIDER Transition et stabilité politique d'un système redistributif 	G 9722	Crédibilité, indépe Une revue de la lit P. AUGERAUD - I
G 9703	D. GOUX - E. MAURIN Train or Pay: Does it Reduce Inequalities to En- courage Firms to Train their Workers?		Les données com Le système interm Passage des donr sectorielles
G 9704	P. GENIER Deux contributions sur dépendance et équité	G 9723	P. AUGERAUD Using Business Ac
G 9705	E. DUGUET - N. IUNG R & D Investment, Patent Life and Patent Value An Econometric Analysis at the Firm Level	G 9724	Accounts: the Frei P. AUGERAUD Les comptes d'ent
G 9706	M. HOUDEBINE - A. TOPIOL-BENSAÏD Les entreprises internationales en France : une analyse à partir de données individuelles		sage aux comptes d'entreprise à la ci <i>paraître</i>
G 9707	M. HOUDEBINE Polarisation des activités et spécialisation des départements en France	G 9801 G 8807	H. MICHAUDON - Présentation du m
G 9708	E. DUGUET - N. GREENAN Le biais technologique : une analyse sur données individuelles	G 9803	Une étude de com pour la France en X RONNFT - S F
G 9709	J.L. BRILLET Analyzing a small French ECM Model		Apports et limites « Real Business C
G 9710	 J.L. BRILLET Formalizing the transition process: scenarios for capital accumulation 	G 9804	C. BARLET - C. D D. ENCAOUA - J. The Commercial S
G 9711	G. FORGEOT - J. GAUTIÉ Insertion professionnelle des jeunes et processus de déclassement		An econometric ar French manufactu

E. DUBOIS High Real Interest Rates: the Consequence of a Saving Investment Disequilibrium or of an in- sufficient Credibility of Monetary Authorities?	U	G 9805	م م ش م
Bilan des activités de la Direction des Études et Synthèses Économiques - 1996 E - Econimi EED	U	G 9806	J. 19
Price Undex Over- Does the French Consumer Price Index Over- state Inflation?	U	G 9807	Sy Bil
 X. BONNET Peut-on mettre en évidence les rigidités à la baisse des salaires nominaux ? 	U	G 9808	₹ö 8
Une etude sur queiques grands pays de l'OCDE N. IUNG - F. RUPPRECHT Productivité de la recherche et rendements	U	G 9809	× Å S
d'échelle dans le secteur pharmaceutique français E. DUGUET - I. KABLA	U	G 9810	щ 🗞 🖻
Appropriation strategy and the motivations to use the patent system in France - An econometric analysis at the firm level	U	G 9811	
L.P. PELÉ - P. RALLE Âge de la retraite : les aspects incitatifs du ré- gime général	C	0010	ge∯ (
ZHANG Yingxiang - SONG Xueqing Lexique macroéconomique français-chinois, chinois-francais	י פ		. هد.
M. HOUDEBINE - J.L. SCHNEIDER Mesurer l'influence de la fiscalité sur la locail- sation des entreprises	פ	G 9813	
A, MOUROUGANE Crédibilité, indépendance et politique monétaire Une revue de la littérature	0 0	G 9814	ല് വ് വ
P. AUGERAUD - L. BRIOT Les données comptables d'entreprises Le système intermédiaire d'entreprises	ט	G 9901	a, p, è, ⊑
Passage des données individuelles aux données sectorielles	U	G 9902	ΰž
P. AUGERAUD - J.E. CHAPRON Using Business Accounts for Compiling National Accounts: the French Experience	U	G 9903	с Ф Ф В
P. AUGERAUD Les comptes d'entreprise par activités - Le pas- sage aux comptes - De la comptabilité	U	G 9904	ы, <u>г</u>
d'entreprise à la comptabilité nationale - A paraître	U	G 9905	щŞА
H. MICHAUDUN - C. FRIGEN I Présentation du modèle AMADEUS	U	G 9906	υZ
J. ACCARDO Une étude de comptabilité générationnelle pour la France en 1996			fra
X. BONNET - S. DUCHÊNE Apports et limites de la modélisation « Real Business Cycles »	U	G 9907	сі. Г.
C. BARLET - C. DUGUET -	U	G 9908	Bil
D. ENCAOUA - J. PRADEL The Commercial Success of Innovations An econometric analysis at the firm level in	U	G 9909	ЧОЙ
French manufacturing	0	G 9910	Ϋ́Ξ

I	iv
P. CAHUC - Ch. GIANELLA - D. GOUIX - A ZII REPREPG	G 9911
Equalizing Wage Differences and Bargaining Power - Evidence form a Panel of French Firms	G 9912
 J. ACCARDO - M. JLASSI La productivité globale des facteurs entre 1975 et 1996 	G 9912 Bis
Bilan des activités de la Direction des Études et Synthèses Économiques - 1997	500 U
A. MOUROUGANE Can a Conservative Governor Conduct an Ac- comodative Monetary Policy?	
X. BONNET - E. DUBOIS - L. FAUVET Asymétrie des inflations relatives et menus costs : tests sur l'inflation française	G 3314
E. DUGUET - N. IUNG Sales and Advertising with Spillovers at the firm level: Estimation of a Dynamic Structural Model on Panel Data	
J. P. BERTHIER Congestion urbaine : un modèle de trafic de pointe à courbe débit-vitesse et demande élastique	G 9917
C. PRIGENT La part des salaires dans la valeur ajoutée : une approche macroéconomique	G 9918
A.Th. AERTS L'évolution de la part des salaires dans la valeur ajoutée en France reflète-t-elle les évolutions individuelles sur la période 1979-1994 ?	G 9919
B. SALANIÉ Guide pratique des séries non-stationnaires	
S. DUCHÊNE - A. JACQUOT Une croissance plus riche en emplois depuis le début de la décennie ? Une analyse en compa- raison internationale	G 2000/02
Ch. COLIN Modélisation des carrières dans Destinie	
Ch. COLIN Évolution de la dispersion des salaires : un essai de prospective par microsimulation	G 2000/03
B. CREPON - N. IUNG Innovation, emploi et performances	G 2000/04
B. CREPON - Ch. GIANELLA Wages inequalities in France 1969-1992 An application of quantile regression techniques	G 2000/05
C. BONNET - R. MAHIEU Microsimulation techniques applied to inter- generational transfers - Pensions in a dynamic framework: the case of France	G 2000/06
 F. ROSENWALD L'impact des contraintes financières dans la dé- cision d'investissement 	G 2000/07
Bilan des activités de la DESE - 1998	20002
J. P. ZOYEM Contrat d'insertion et sortie du RMI Évaluation des effets d'une politique sociale	G 2000/06
Ch. COLIN - FI. LEGROS - R. MAHIEU Bilans contributifs comparés des régimes de retraite du secteur privé et de la fonction publique	

G. LAROQUE - B. SALANIÉ Une décomposition du non-emploi en France	 B. SALANIÉ Une maquette analytique de long terme du marché du travail 	Ch. GIANELLA Une estimation de l'élasticité de l'emploi peu qualifié à son coût	Division « Redistribution et Politiques Sociales » Le modèle de microsimulation dynamique DESTINIE	E. DUGUET Macro-commandes SAS pour l'économétrie des panels et des variables qualitatives	R. DUHAUTOIS Évolution des flux d'emplois en France entre 1990 et 1996 : une étude empirique à partir du fichier des bénéfices réels normaux (BRN)	J.Y. FOURNIER Extraction du cycle des affaires : la méthode de Baxter et King	B. CRÉPON - R. DESPLATZ - J. MAIRESSE Estimating price cost margins, scale economies and workers' bargatining power at the firm level	Ch. GIANELLA - Ph. LAGARDE Productivity of hours in the aggregate production function: an evaluation on a panel of French firms from the manufacturing sector	S. AUDRIC - P. GIVORD - C. PROST Évolution de l'emploi et des coûts par quali- fication entre 1982 et 1996	R. MAHIEU Les déterminants des dépenses de santé : une approche macroéconomique	C. ALLARD-PRIGENT - H. GUILMEAU - A. QUINET The real exchange rate as the relative price of nontrables in terms of tradables: theoretical investigation and empirical study on French data	 JY. FOURNIER L'approximation du filtre passe-bande proposée par Christiano et Filzgerald 	Bilan des activités de la DESE - 1999	B. CREPON - F. ROSENWALD Investissement et contraintes de financement : le poids du cycle Une estimation sur données françaises	A. FLIPO Les comportements matrimoniaux de fait	R. MAHIEU - B. SÉDILLOT Microsimulations of the retirement decision: a supply side approach	 C. AUDENIS - C. PROST Déficit conjoncturel : une prise en compte des conjonctures passées 	R. MAHIEU - B. SÉDILLOT Équivalent patrimonial de la rente et souscription de retraite complémentaire
9911	9912	9912 Bis	9913	9914	9915	9916	9917	9918	9919	2000/01	2000/02	2000/03	2000/04	2000/05	2000/06	2000/07	2000/08	2000/09

La reprise de la croissance de l'emploi profite-t-elle aussi aux non-diplômés ? Le canal du crédit fonctionne-t-il en Europe ? Une étude de l'hétérogénélité des comportements d'investissement à partir de données de bilan Les dispositifs d'acquisition à titre facultatif Comparaison des salaires des secteurs public et between capital, skilled and less skilled workers: an analysis at the firm level in the French Modelling demographic behaviours in the French Festing the augmented Solow growth model: An Fiscalité, coût d'usage du capital et demande de dispositifs Une méthodologie de comparaison entre consommations intermédiaires de source fiscale Computerization in France: an evaluation based Évolution et répartition du surplus de productivité Temps partiel féminin et incitations financières à facteurs : une analyse sur données individuelles Ralentissement de l'investissement : petites ou Départ à la retraite, irréversibilité et incertitude Informatisation en France : une évaluation à grandes entreprises ? industrie ou tertiaire ? empirical reassessment using panel data de charges sociales sur les bas salaires Substitution and complementarity Bilan des activités de la DESE - 2000 The new economy and the measure des La nouvelle économie et la mesure -ocal unemployment and wages R. CONVENEVOLE - S. PISANI P. BISCOURP - Ch. GIANELLA C. AUDENIS - P. BISCOURP -J.-P. BERTHIER - C. JAULENT partir de données individuelles d'annuités viagères de retraite B. CRÉPON - Ch. GIANELLA effets B. CRÉPON - R. DESPLATZ **N. FOURCADE - O. LOISEL** G. LAROQUE - B. SALANIÉ B. CREPON - Th. HECKEL on individual company data R. MAHIEU - B. SÉDILLOT et de comptabilité nationale A. BEAUDU - Th. HECKEL de la croissance du PIB ment of GDP growth manufacturing industry **BRAUN-LEMAIRE** des J. Ph. GAUDEMET I. ROBERT-BOBEE J.-Y. FOURNIER R. DUHAUTOIS Ch. GIANELLA F. LEQUILLER d'allégements Évaluation S. AUDRIC agrégées emploi orivé G 2000/10 G2001/13 G 2000/11 G2000/12 G2000/13 G2001/02 G2001/03 G2001/05 G2001/06 G2001/08 G2001/09 G2001/10 G2001/11 G2001/12 G2001/14 G2001/01 G2001/04 G2001/07

des Retropolation of the investment series (GFCF) and estimation of fixed capital stocks on the ESA-95 basis for the French balance sheets d'assurance vieillesse à l'aide du modèle de microsimulation DESTINIE Les comportements démographiques dans le modèle de microsimulation Destinie - Une microsimulation model Destinie: An analysis of a réduction des taux d'activité aux âges Existe-t-il une asymétrie dans la transmission du Les statistiques internationales de PIB par habitant en standard de pouvoir d'achat: une a cessation d'activité au sein des couples : y a-t-Rétropolation des séries de FBCF et calcul du capital fixe en SEC-95 dans les comptes How do firms respond to cheaper computers? Microeconometric evidence for France based on ę l'information et de la communication sur l'économie française - un bouclage macro-J. BARDAJI - B. SÉDILLOT - E. WALRAET Évaluation de trois réformes du Régime Général Réflexions sur les différentes notions de volume dans les comptes nationaux : comptes aux prix d'une année fixe ou aux prix de l'année Les soldes d'opinion résument-ils au mieux les éponses des entreprises aux enquêtes de comparaison des estimations issues des anquêtes Jeunes et Carrières 1997 et Histoire -a dynamique des bas revenus : une analyse des P. BISCOURP - B. CRÉPON - T. HECKEL - N. F. MAGNIEN - J.-L. TAVERNIER - D. THESMAR C. AUDENIS - P. BISCOURP - N. RIEDINGER C. AUDENIS - J. DEROYON - N. FOURCADE calendrier l'impact des nouvelles technologies Diagnostic sur la pauvreté et calendrier revenus : le cas du "Panel européen prix du brut aux prix des carburants ? Bilan des activités de la DESE - 2001 extrêmes, une spécificité française ? Prévisions d'inflation pour la France uture change in completed fertility a production function approach J.-Y. FOURNIER - P. GIVORD l interdépendance des choix ? B. SÉDILLOT - E. WALRAET précédente, séries chaînées entrées-sorties de pauvreté analyse des résultats nationaux français ROBERT-BOBÉE .-P. BERTHIER G. BRILHAULT ⁻amiliale 1999 conjoncture ? I.-P. ZOYEM I.-P. ZOYEM REDINGER économique nénages » HILD F. HILD G2001/15 G2002/10 G2002/12 32001/16 G2001/17 G2002/02 G2002/03 G2002/05 G2002/06 G2002/07 G2002/08 G2002/09 G2002/01 G2002/04 G2002/11

G20 .2 reur-un meure en évidence l'existence de rigidités à la baisse des salaires à partir de données induelles ? Le cas de la France à la Création d'emplois, destruction d'emplois et internationalisation des entreprises industrielles La compétitivité exprimée dans les enquêtes trimestrielles sur la situation et les perspectives Réduction du temps de travail et tensions sur les · Analyse de la redistribution intragénérationnelle dans le système de retraite des salariés du privé · Intragenerational distributional analysis in the french private sector pension scheme - A Offre de travail féminine et garde des jeunes Le coût de dépollution atmosphérique pour les entreprises françaises : Une estimation à partir de données individuelles : une Age, salaire et productivité La productivité des salariés décline-t-elle en fin H. BARON - P.O. BEFFY - N. FOURCADE - R. au Patrimoine des ménages, dynamique d'allocation Présence syndicale dans les firmes : quel impact sur les inégalités salariales entre les hommes et F. MAUREL - S. GREGOIR Les indices de compétitivité des pays : inter-La situation des salariés de plus de cinquante P.-O. BEFFY - X. BONNET - M. DARRACQrançaises : une analyse sur la période 1986-P. CHONE - D. LE BLANC - I. ROBERT-BOBEE Évolutions démographiques et croissance : u projection macro-économique à l'horizon 2020 Le ralentissement de la productivité du travail MZE: a small macro-model for the euro area N. FOURCADE - S. GREGOIR - N. LAÏB -Bilan des activités de la DESE - 2002 Une approche par microsimulation et comportement de consommation P. BISCOURP - N. FOURCADE P. BISCOURP et F. KRAMARZ P.-O. BEFFY - J. DEROYON -P.-O. BEFFY - B. MONFORT E. WALRAET - A. VINCENT N. RIEDINGER - E.HAUVY P. AUBERT - M. LECLAIR P. AUBERT - B. CRÉPON microsimulation approach ans dans le secteur privé PARIES - B. MONFORT M. LECLAIR - P. PETIT cours des années 1990 facteurs de production prétation et limites fin des années 90 dans l'industrie **3. MONFORT** les femmes ? M. LECLAIR de carrière ? P. AUBERT MAHIEL enfants 1992 G2002/13 32002/15 32002/16 G2003/10 G2002/14 G2003/01 32003/02 G2003/03 G2003/04 G2003/05 G2003/06 G2003/07 32003/08 G2003/09 G2003/11 G2004/01

M. DUÉE - C. REBILLARD La dépendance des personnes âgées : une projection à long terme	 S. RASPILLER - N. RIEDINGER Régulation environnementale et choix de localisation des groupes français 	A. NABOULET - S. RASPILLER Les déterminants de la décision d'investir : une approche par les perceptions subjectives des firmes	N. RAGACHE La déclaration des enfants par les couples non mariés est-elle fiscalement optimale ?	M. DUÉE L'impact du chômage des parents sur le devenir scolaire des enfants	P. AUBERT - E. CAROLI - M. ROGER New Technologies, Workplace Organisation and the Age Structure of the Workforce: Firm-Level Evidence	E. DUGUET - C. LELARGE Les brevets accroissent-ils les incitations privées à innover ? Un examen microéconométrique	S. RASPILLER - P. SILLARD Affiliating versus Subcontracting: the Case of Multinationals	J. BOISSINOT - C. L'ANGEVIN - B. MONFORT Public Debt Sustainability: Some Results on the French Case	S. ANANIAN - P. AUBERT Travailleurs âgés, nouvelles technologies et changements organisationnels : un réexamen à partir de l'enquête « REPONSE »	X. BONNET - H. PONCET Structures de revenus et propensions différentes à consommer - Vers une équation de oonsommation des ménages plus robuste en prévision pour la France	 C. PICART Évaluer la rentabilité des sociétés non financières 	J. BARDAJI - B. SÉDILLOT - E. WALRAET Les retraites du secteur public: projections à l'horizon 2040 à l'aide du modèle de microsimulation DESTINIE	 S. BUFFETEAU - P. GODEFROY Conditions de départ en retraite selon l'âge de fin d'études : analyse prospective pour les générations 1945 à1974 	C. AFSA - S. BUFFETEAU L'évolution de l'activité féminine en France : une approche par pseudo-panel	P. AUBERT - P. SILLARD Délocalisations et réductions d'effectifs dans l'industrie française	M. LECLAIR - S. ROUX Mesure et utilisation des emplois instables dans les entreprises	C. L'ANGEVIN - S. SERRAVALLE Performances à l'exportation de la France et de l'Allemagne - Une analyse par secteur et destination géographique
004/02	004/03	004/04	004/05	004/06	004/07	004/08	004/09	004/10	004/11	004/12	004/13	004/14	005/01	005/02	005/03	005/04	005/05

G2006/15 G2006/10 G2006/12 G2006/13 32006/14 G2007/09 G2007/10 G2006/07 G2006/08 G2006/09 G2006/11 G2007/01 G2007/02 G2007/03 G2007/04 G2007/05 G2007/06 32007/07 G2007/08 Bilan des activités de la Direction des Études et Le rendement apparent de la formation continue dans les entreprises : effets sur la productivité et ites des propriétés G. BIAU - O. BIAU - L. ROUVIERE Nonparametric Forecasting of the Manufacturing Output Growth with Firm-level Survey Data Éducation et croissance en France et dans un Un indicateur des révisions dans l'enquête de conjoncture sur les investissements dans A two-states Markov-switching model of inflation in France and the USA: credible target VS de premiers La modélisation des comportements démogra-phiques dans le modèle de microsimulation Disparités de retraite de droit direct entre hommes et femmes : quelles évolutions ? Exemple de la prévision de la production elements Étude de simulation sur la participation versée fiscale : principaux enseignechocs C. BONNET - S. BUFFETEAU - P. GODEFROY O. BIAU - H. ERKEL-ROUSSE - N. FERRARI Réponses individuelles aux enquêtes conjoncture et prévision macroéconomiques : apport d'un modèle « qualifiés/peu qualifiés » what do macroeconometric models predict? implicites Demographic change and unemployment: des P. AUBERT - D. BLANCHET - D. BLAU The labour market after age 50: some P. AUBERT - B. CRÉPON - P. ZAMORA Prévoir l'investissement des entreprises et chiffrages, P. DOURGNON - P. POLLET L'enquête SHARE : présentation et réactions à Chômage et boucle prix-salaires : of a Franco-American comparison J.-F. OUVRARD - R. RATHELOT D. BLANCHET - J.-F. OUVRARD La concurrence fiscale : princ ments de l'analyse économique D. BLANCHET - T. DEBRAND -Synthèses Économiques - 2004 aux salariés par les entreprises d'engagements P.-O. BEFFY - C. L'ANGEVIN ésultats de l'édition française panel de 21 pays de l'OCDE systèmes de retraite : C. L'ANGEVIN - N. LAÏB Les gazelles en France démographiques types H. RAOUI - S. ROUX S. RASPILLER manufacturière nflation spiral N. FERRARI analytiques Indicateurs es salaires C. PICART M. DUÉE 'industrie. B. HEITZ DESTINE G2005/06 G2005/10 G2005/12 G2005/13 G2005/14 G2005/15 G2005/16 G2006/05 G2006/06 G2005/08 G2005/09 G2005/11 G2006/02 G2006/03 G2005/07 G2006/01 G2006/04

:17

The Changing response to oil price shocks in France: a DSGE type approach Une méthodologie d'évaluation comptable du coût du capital des entreprises françaises : 1984entre et -es rendements non monétaires de l'éducation : de satisfaction : -e rôle des conditions de travail dans les P. SILLARD - C. L'ANGEVIN - S. SERRAVALLE Performances comparées à l'exportation de la -'estimation d'un coût implicite de la pénibilité du les plants location Diversification, recentrage et poids des activités Aspiration à la retraite, santé et satisfaction au Quel impact des variations du prix du pétrole sur -lux d'emploi et de main-d'œuvre en France : un for -es entreprises (industrielles) françaises sontsalariales dans Evidence from the French Real Estate Market Consumption over the Life Cycle: Facts patronales ⁻aut-il pondérer les réponses individuelles ? relation démocratisation Collateral Value and Corporate Investment de support dans les groupes (1993-2000) T. CHANEY - D. SRAER - D. THESMAR ⁻rance et de ses principaux partenaires ravail : une comparaison européenne 'enseignement supérieur en France Jne analyse structurelle sur 12 ans les variables de elles à la frontière technologique ? <u>a</u> Ч cotisations ravail chez les travailleurs âgés 'exemple de la durée du travail D. BLANCHET - T. DEBRAND R. RATHELOT - P. SILLARD The impact of local taxes þ **GONZALEZ - C. PICART** M. BARLET - L. CRUSSON productivité et inégalités X. BOUTIN - S. QUANTIN V. ALBOUY - L. LEQUIEN a croissance française ? V. ALBOUY - C. TAVAN O. BIAU - N. FERRARI absences pour maladie Jne réinterprétation C. AFSA - P. GIVORD T. LE BARBANCHON A. KOUBI - S. ROUX ŧ dynamique salariale lhéorie de l'opinion qe le cas de la santé I. BOISSINOT Allègements C. LELARGE Massification nterpréter entreprises D. SRAER C. PICART 'éexamen C. AFSA C. AFSA decision -rance 2002

viii effets sur créations Évolutions démographiques et déformation du cycle de vie active : quelles relations ? and évaluation dans le beu Économies d'agglomération et productivité des entreprises : estimation sur données individuelles principaux Internationalisation et flux d'emplois : que dit une approche comptable ? Evidence from a French Loan Guarantee géographique dans l'industrie Who is confronted to insecure labor market histories? Some evidence based on the French Social Security and Well-Being of the Elderly: the Analyser les composantes du bien-être et de son données A Monthly Indicator of the French Business manufacturière et dans les services en France mécanismes et retour sur l'expérience française Microsimuler le marché du travail : un prototype Entrepreneurship and Credit Constraints effets Do Business Tendency Surveys in Industry Services Help in Forecasting GDP Growth? M. BARLET - D. BLANCHET - L. CRUSSON françaises : rentables mais The Effects of Retail Regulations on Prices Le partage de la valeur ajoutée en France, 1949-2007 C. LELARGE - D. SRAER - D. THESMAR une approche par un indicateur en continu M. BEFFY - É. COUDIN - R. RATHELOT M. BARLET - A. BRIANT - L. CRUSSON P. BISCOURP - X. BOUTIN - T. VERGÉ -aurent CLAVEL - Christelle MINODIER Are Prices Really Affected by Mergers? sur quels es projections démographiques : A Real-Time Analysis on French Data D. BLANCHET - F. TOUTLEMONDE H. ERKEL-ROUSSE - C. MINODIER les cadre du modèle causal de Rubin empirique D. BLANCHET - F. LE GALLO Evidence form the Loi Galland M. BARLET - D. BLANCHET -Zones Franches Urbaines : R. RATHELOT - P. SILLARD Aléa moral en santé: une Y. BARBESOL - A. BRIANT et M. ROGER - E. WALRAET V. ALBOUY - B. CRÉPON X. BOUTIN - L. JANIN labor market transition . LE BARBANCHON salarié d'établissements ? approche P.-A. PIONNIER Case of France Concentration C. PICART Les PME dynamiques individuelles françaises évolution Program C. AFSA l'emploi Climate Jne G2007/11 G2007/12 32008/10 G2008/12 G2008/13 G2008/02 32008/03 G2008/04 32008/05 32008/06 G2008/07 32008/08 G2008/09 G2008/11 32009/02 32009/03 G2008/01 G2009/01

N. CECI-RENAUD - P.-A. CHEVALIER Les seuils de 10, 20 et 50 salariés : impact sur la taille des entreprises françaises G. LALANNE - P.-A. PIONNIER - O. SIMON Le partage des fruits de la croissance de 1950 à 2008 : une approche par les comptes de surplus Adjusted net savings and other approaches to sustainability: some theoretical background Une décomposition du compte des ménages de la comptabilité nationale par catégorie de ð V. BELLAMY - G. CONSALES - M. FESSEAU -S. LE LAIDIER - É. RAYNAUD Detecting Economic Regimes in France: a Qualitative Markov-Switching Indicator Using par The Effect of the Uruguay round on the Intensive and Extensive Margins of Trade Avantages comparés des séries des premières valeurs publiées et des séries des valeurs L'IPC, miroir de l'évolution du coût de la vie en France ? Ce qu'apporte l'analyse des courbes -es contrats temporaires : trappe ou marchepied ⁻aut-il pondérer ?... Ou l'éternelle question de l'économètre confronté à des données d'enquête S. QUANTIN - S. RASPILLER - S. SERRAVALLE fiscalité et prix de Élasticités-prix des consommations énergétiques Prix du pétrole et croissance potentielle à long Discrimination à l'embauche : comment exploiter révisées - Un exercice de prévision en temps réel D. BLANCHET - J. LE CACHEUX - V. MARCUS Tome 1 – Version avec volumes à prix constants Commerce intragroupe, fiscalité et prix d transferts : une analyse sur données françaises de la croissance trimestrielle du PIB en France Health Expenditure Models: a Comparison -e modèle MÉSANGE réestimé en base 2000 FOUGÈRE la comptabilité nationale par catégorie Partage de la valeur ajoutée, approche G. LALANNE - E. POULIQUEN - O. SIMON Y. BARBESOL - P. GIVORD - S. QUANTIN V. ALBOUY - L. DAVEZIES - T. DEBRAND L. DAVEZIES - X. D'HAULTFOEUILLE Five Specifications using Panel Data . ص données microéconomiques les procédures de testing ? M.-É. CLERC - É. COUDIN M. CLERC - V. MARCUS I. BUONO - G. LALANNE P. GIVORD - L. WILNER J. BARDAJI - F. TALLET Mixed Frequency Data AEBERHARDT C. KLEIN - O. SIMON vers l'emploi stable ? ménage en 2003 R. RATHELOT des ménages C. MINODIER d'Engel terme ż G2009/12 G2009/04 G2009/06 G2009/09 G2009/10 G2009/13 G2009/14 G2009/15 G2010/02 G2010/03 G2010/05 G2009/05 G2009/07 32009/08 G2009/11 G2010/04 G2010/01

V. LAPÈGUE - O. MONSO M. ROGER - M. WASMER P. FÉVRIER - L. WILNER Report: A few illustrations matters: C. LOUVOT-RUNAVOT C. MARBOT - D. ROY à domicile en 2007 Urbaines program Heterogeneity rois modèles X. BOUTIN mooc Firms 2008 G2011/04 G2011/06 G2011/08 G2011/09 G2011/10 G2011/11 G2011/12 G2011/13 G2011/14 G2011/15 G2011/16 G2011/17 G2012/02 G2012/03 G2011/05 G2011/07 G2012/01 L'apport des méthodes à noyaux pour mesurer la concentration géographique - Application à la concentration des immigrés en France de 1968 à National Origin Differences in Wages and Hierarchical Positions - Evidence on French Full-Time Male Workers from a matched Employer-<u>m</u> Modèles à effets fixes, à effets aléatoires, modèles mixtes ou multi-niveaux : propriétés et mises en œuvre des modélisations de Les trajectoires professionnelles en début de vie Méthodes économétriques pour l'évaluation de M. BARATON - M. BEFFY - D. FOUGÈRE Une évaluation de l'effet de la réforme de 2003 sur les départs en retraite - Le cas des Le modèle de microsimulation Destinie 2 : principales caractéristiques et premiers résultats M. BARLET - L. CRUSSON - S. DUPUCH -Public-private wage gaps: is civil-servant human capital sector-specific? es marchés du travail français et américain face aux chocs conjoncturels des années 1986 à 2007 : une modélisation DSGE D. BLANCHET - S. BUFFETEAU - E. CRENNER Conditional Logit with one Binary Covariate: Link Une évaluation de la réduction d'impôt pour Des services échangés aux services échangeables : une application sur données françaises T. LE BARBANCHON - B. OURLIAC - O. SIMON 'hétérogénéité dans le cas de données groupées Quelle croissance de moyen terme après crise ? P.-Y. CABANNES - H. ERKEL-ROUSSE active : quel impact des contrats temporaires ? Tome 2 - Version avec volumes à prix chaînés G. LALANNE - O. MONSO - E. POULIQUEN -e modèle Mésange réestimé en base 2000 à la baisse des tarifs douaniers étrangers E. POULIQUEN - M. BEFFY - M. GAINI between the Static and Dynamic Cases Le bloc retraites du modèle Destinie 2 : La réaction des entreprises françaises enseignants du second degré public P.-Y. CABANNES - V. LAPÈGUE -R. AEBERHARDT - L. DAVEZIES R. AEBERHARDT - J. POUGET D. BLANCHET - E. CRENNER 'emploi de salariés à domicile R. RATHELOT - P. SILLARD M. BEFFY - T. KAMIONKA I. BUONO - G. LALANNE S. BLASCO - P. GIVORD guide de l'utilisateur politiques publiques Employee Dataset -. DAVEZIES S. LE MINEZ C. MARBOT P. GIVORD =. PUECH 666 G2010/06 G2010/10 G2010/12 G2010/13 G2010/14 G2010/15 G2010/16 G2010/17 G2010/18 G2011/02 G2011/03 G2010/08 G2010/09 G2010/11 G2011/01 G2010/07

Ľ.

M. BARLET - M. CLERC - M. GARNEO -La tertiarisation de l'économie française et le ralentissement de la productivité entre 1978 et labour productivity De la crise financière à la crise économique L'impact des perturbations financières de 2007 et Recommendations of the Stiglitz-Sen-Fitoussi Projeter l'impact des réformes des retraites sur 'activité des 55 ans et plus : une comparaison de L'évaluation de l'activité dissimulée des entre-prises sur la base des contrôles fiscaux et son Évaluation de la transformation de la réduction d'impôt en crédit d'impôt pour l'emploi de salariés Place-based tax exemptions and displacement effects: An evaluation of the Zones Franches The Environmental Effect of Green Taxation: the -a nouvelle version du modèle MZE, modèle Learning, Incomplete Contracts and Export Dynamics: Theory and Evidence form French A Long-Term Evaluation of the First Generation Correctly Expect Price 2008 sur la croissance de sept pays industrialisés ^oolitique salariale et performance des entreprises -es inégalités entre générations depuis le baby-R. AEBERHARDT - I. BUONO - H. FADINGER GIVORD M.-É. CLERC - O. MONSO - E. POULIQUEN M. BACHELET - M. BEFFY - D. BLANCHET P. GIVORD - R. RATHELOT - P. SILLARD P. GIVORD - S. QUANTIN - C. TREVIEN Nage inequalities in France 1976-2004: macroéconométrique pour la zone euro Do Consumers Correctly ⊏∧עיי Reductions? Testing Dynamic Behavior P. CHARNOZ - É. COUDIN - M. GAINI M. CLERC - M. GAINI - D. BLANCHET I.-C. BRICONGNE - J.-M. FOURNIER of the French Urban Enterprise Zones nsertion dans les comptes nationaux Restrictive Fiscal Policies in Europe: D'HAULTFOEUILLE - P. Case of the French "Bonus/Malus" N. CECI-RENAUD - V. COTTET differentiated by age and skills a quantile regression analysis C. KERDRAIN - V. LAPÈGUE A. SCHREIBER - A. VICARD V. LAPÈGUE - V. MARCUS What are the Likely Effects?

x	G2013/	G2013/	G2013/	G2013/	G2013/	G2013/	G2013/	12002	G2014/	G2014/	G2014/	G2014/	G2014/	G2014/	G2014/	G2014/	G2014/	
	M. GAINI - A. LEDUC - A. VICARD School as a shelter? School leaving-age and the business cycle in France	M. GAINI - A. LEDUC - A. VICARD A scarred generation? French evidence on young people entering into a tough labour market	P. AUBERT - M. BACHELET Disparités de montant de pension et redistribution dans le système de retraite français	R. AEBERHARDT - P GIVORD - C. MARBOT Spillover Effect of the Minimum Wage in France: An Unconditional Quantile Regression Approach	 A. EIDELMAN - F. LANGUMIER - A. VICARD Prélèvements obligatoires reposant sur les ménages : des canaux redistributifs différents en 1990 et 2010 	O. BARGAIN - A. VICARD Le RMI et son successeur le RSA découragent- ils certains jeunes de travailler ? Une analyse sur les jeunes autour de 25 ans	C. MARBOT - D. ROY Projections du coût de l'APA et des caractéristiques de ses bénéficiaires à l'horizon 2040 à l'aide du modèle Destinie	 A. MAUROUX Le crédit d'impôt dédié au développement durable : une évaluation économétrique 	V. COTTET - S. QUANTIN - V. RÉGNIER Coût du travail et allègements de charges : une estimation au niveau établissement de 1996 à 2008	X. D'HAULTFOEUILLE - P. FÉ'URIER - L. WILNER Demand Estimation in the Presence of Revenue Management	D. BLANCHET - S. LE MINEZ Joint macro/micro evaluations of accrued-to-date pension liabilities: an application to French reforms	T. DEROYON - A. MONTAUT - P-A PIONNIER Utilisation rétrospective de l'enquête Emploi à une fréquence mensuelle : apport d'une modélisation espace-état	C. TREVIEN Habiter en HLM: quel avantage monétaire et quel impact sur les conditions de logement ?	A. POISSONNIER Temporal disaggregation of stock variables - The Chow-Lin method extended to dynamic models	P. GIVORD - C. MARBOT Does the cost of child care affect female labor market participation? An evaluation of a French reform of childcare subsidies	G. LAME - M. LEQUIEN - PA. PIONNIER Interpretation and limits of sustainability tests in public finance	C. BELLEGO - V. DORTET-BERNADET La participation aux pôles de compétitivité : quelle incidence sur les dépenses de R&D et l'activité des PME et ETI ?	
	G2012/04	G2012/05	G2012/06	G2012/07	G2012/08	G2012/09	G2012/10	G2012/11	G2012/12	G2012/13	G2012/14	G2013/01- F1301	G2013/02- F1302	G2013/03	G2013/04	G2013/05	G2013/06	

S. GEORGES-KOT Annual and lifetime incidence of the value-added	tax in France M. POULHÈS Are Enterprise Zones Benefits Capitalized into Commercial Property Values? The French Case	JB. BERNARD - O. LAFFÉTER Effet de l'activité et des prix sur le revenu salarial des différentes catégories socioprofessionnelles	C. GEAY - M. KOUBI - G de LAGASNERIE Projections des dépenses de soins de ville, construction d'un module pour Destinie	J. BARDAJI - JC. BRICONGNE - B. CAMPAGNE - G. GAULIER Compared performances of French companies on the domestic and foreign markets	C. BELLÉGO - R. DE NJJS The redistributive effect of online piracy on the box office performance of American movies in foreign markets	JB. BERNARD - L. BERTHET French households financial wealth: which changes in 20 years?	Penetre sur cour ou Chamore avec vue ? Les prix hédoniques de l'immobilier parisien B. GARBINT - S. EGORGES-KOT Time to emeil the mose? Stie vuerent the		Multi-Plant Businesses: E of the French High-Speed	BONNET - B. GARBINTI - A. SO ander Inequality after Divorce: Th arital Specialization - Evidence f iministrative Database	D. BLANCHET - E. CAROLI - C. PROST - M. ROGER Health capacity to work at older ages in France B. CAMPAGNE - A. POISSONNIER	MELEZE: A DSGE model for France within the Euro Area B. CAMPAGNE - A. POISSONNIER	Larrer curves and riscal murphiers: lessons from Mélèze model B. CAMPAGNE - A. POISSONNIER Structural reforms in DSGE models: a case for senstitvith analvese	Y. DUBOIS et M. KOUBI Y. DUBOIS et M. KOUBI Relèvement de l'âge de départ à la retraite : quel impact sur l'activité des séniors de la réforme des retraites de 2010 7	A. NAOUAS - M. ORAND - I. SLIMANI HOUTI Les entreprises employant des salarrés au Smic : quelles caractéristiques et quelle rentabilité ?	T. BLANCHET - Y. DUBOIS - A. MARINO - M. ROGER Patrimoine privé et retraite en France	M. PAK - A. POISSONNIER Accounting for technology, trade and final
G2015/12	G2015/13	G2015/14	G2015/15	G2015/16	G2015/17	G2015/18 G2015/19	G2016/01	G2016/02		G2016/03	G2016/04 G2016/05	G2016/06	G2016/07	G2016/08	G2016/09	G2016/10	G2016/11
B. GARBINTI - P. LAMARCHE Les hauts revenus épargnent-ils davantage ?	D. AUDENAERT - J. BARDAJI - R. LARDEUX - M. ORAND - M. SICSIC Wage Resilience in France since the Great Recession	F. ARNAUD - J. BOUSSARD - A. POISSONNIER - H. SOUAL Computing additive contributions to growth and other issues for chain-linked cuartery accreates	H. FRAISSE - F. KRAMARZ - C. PROST Labor Disputes and Job Flows	P. GIVORD - C. GRISLAIN-LETRÉMY - H. NAEGELE How does fuel taxation impact new car purchases? An evaluation using French consumer-laved dataset	P. AUBERT - S. RABATÉ P. AUBERT - S. RABATÉ Durée passée en carrière et durée de vie en retraite : quel partage des gains d'espérance de	Act A. POISSONNIER The walking dead Euler equation Addressing a challenge to monetary policy models	Y. DUBOIS - A. MARINO Indicateurs de rendement du système de retraite français	T. MAYER - C. TREVIEN The impacts of Urban Public Transportation: Evidence from the Paris Region	S.T. LY - A. RIEGERT Measuring Social Environment Mobility	M. A. BEN HALIMA - V. HYAFIL-SOLELHAC M. KOUBI - C. REGAERT Quel est l'impact du système d'indemnisation maladie sur la durée des arrêts de travail pour maladie?	Y. DUBOIS - A. MARINO Disparités de rendement du système de retraite dans le secteur privé : approches intergénéra- tionnelle et intradénérationnelle	B. CAMPAGNE - V. ALHENC-GELAS - JB. BERNARD No evidence of financial accelerator in France	Q. LAFFÉTER - M. PAK Élasticités des recettes fiscales au cycle économique : étude de trois impôts sur la période 1979-2013 en France	 JM. DAUSSIN-BENICHOU, S. IDMACHICHE, A. LEDUC et E. POULIQUEN Les déterminants de l'attractivité de la fonction publique de l'État 	P. AUBERT La modulation du montant de pension selon la durée de carritére et l'âge de la retraite : quelles disparités entre assurés ?	V. DORTET-BERNADET - M. SICSIC Effet des aides publiques sur l'emploi en R&D dans les petites entrenises	
G2014/10	G2014/11	G2014/12	G2014/13	G2014/14	G2014/15	G2015/01	G2015/02	G2015/03	G2015/04	G2015/05	G2015/06	G2015/07	G2015/08	G2015/09	G2015/10	G2015/11	

G2	G2	G2	G2	(G2 G2	(62 63	G2 G2	ц Т	G2 G2		E 1	G2	G2	G2
consumption in employment: an Input-Output	D.FOUGÈRE -E. GAUTIER - S. ROUX D.FOUGÈRE - E. GAUTIER - S. ROUX Understanding Wage Floor Setting in Industry- Level Agreements: Evidence from France	Y. DUBOIS - M. KOUBI Règles d'indexation des pensions et sensibilité des dépenses de retraites à la croissance économique et aux chocs démographiques	A. CAZENAVE-LACROUTZ - F. GODET L'espérance de vie en retraite sans incapacité sévère des générations nées entre 1960 et 1990 : une projection à partir du modèle Destinie	J. BARDAJI - B. CAMPAGNE - MB. KHDER - Q. LAFFÉTER - O. SIMON (Incea)	(misce) PE. DUFERNEZ - C. ELEZAAR - PEBLANC - E. MASSON - H. PARTOUCHE (DG-Trésor) Le modèle macroéconométrique Mésange : réestimation et nouveautés	J. BOUSSARD - B. CAMPAGNE Fiscal Policy Coordination in a Monetary Union at the Zero-Lower-Bound	A. CAZENAVE-LACROUTZ - A. GODZINSKI Effects of the one-day waiting period for sick leave on health-related absences in the French central civil service	P. CHARNOZ - M. ORAND Qualification, progrès technique et marchés du travail locaux en France, 1990-2011	K. MILIN Modélisation de l'inflation en France par une approche macrosectorielle	CM. CHEVALIER - R. LARDEUX Homeownership and labor market outcomes: disentangling externality and composition effects	P. BEAUMONT Time is Money: Cash-Flow Risk and Export Market Behavior	S. ROUX - F. SAVIGNAC SMEs' financing: Divergence across Euro area countries?	CM. CHEVALIER - A. LUCIANI Computerization, labor productivity and employment: impacts across industries vary with technological level	R.MONIN - M. SUAREZ CASTILLO L'effet du CICE sur les prix : une double analyse sur données sectorielles et individuelles	R. LARDEUX Who Understands The French Income Tax? Bunching Where Tax Liabilities Start
	G2017/01	G2017/02	G2017/03	G2017/04		G2017/05	G2017/06	G2017/07	G2017/08	G2017/09	G2017/10	G2018/01	G2018/02	G2018/03	G2018/04

xii

х.

	_
:	Ξ
	×

ΠY	P. PORA – L. WILNER Child Penalties and Financial Incentives: Exploiting Variation along the Wage Distribution	E. GAUTIER – S. ROUX – M. SUAREZ CASTILLO Do Minimum Wages make Wages more Rigid ? Evidence from French Micro Data	M. ANDRÉ – A. SIREYJOL Imposition des couples et des familles : effets budgétaires et redistributifs de l'impôt sur le revenu	K. MOHKAM – O. SIMON L'empreinte matière de l'économie française : une analyse par matière et catégorie de produits	 S. BUNEL – B. HADJIBEYLI Évaluation du crédit d'impôt innovation 	C. BONNET – F. GODET – A. SOLAZ Gendered economic determinants of couple formation over 50 in France	P. GIVORD – M. SUAREZ CASTILLO Excellence for all? Heterogeneity in high schools' value-added	G. CETTE – L. KOEHL – T. PHILIPPON	ē i	J. SILHUL – B. VEN IELUU Les zones d'intervention prioritaire reflètent- elles des écarts de pratiques des médecins généralistes ?	B. BOUCHTENIK – R. LARDEUX	The Take-Up of Unemployment Benefit Extensions	M. GERMAIN	A Welfare Based Estimate of "Real Feel GDP" for Europe and the USA	J. BOUSSARD – R. LEE	P. PORA	Keep Working and Spend Less? Collective Childcare and Parental Earnings in France	R. MONIN – M. SUAREZ CASTILLO	Product Switching, market power and distance to core competency	L. WILNER	How do citizens perceive centralization reforms? Evidence from the merger of French regions	L. WILNER The persistence of subjective well-being: permanent happiness, transitory misery?
	G2019/08	G2019/09	G2019/10	G2019/11	G2019/12	G2019/13	G2019/14	G2019/15	10,0000	62020/01	G2020/02		G2020/03		G2020/04	G2020/05		G2020/06		G2020/07		G2020/08

 G2020/09 J-M. GERMAIN - T. LELLOUCH
 Comptabilité économique de la soutenabilité climatique
 G2020/10 A. BAUER - M. ROTEMBERG
 Tax avoidance in French Firms: Evidence from the Introduction of a Tax Notch
 G2020/11 P. AGHION - A. BERGEAUD
 M. LEQUIEN - M. J. MELITZ
 The Heterogeneous Impact of Market Size on Innovation: Evidence from French Firm-Level Exports
 G2020/12 L. GALIANA - B. SAKAROVITCH
 F. SÉMÉCURBE - Z. SMOREDA
 Residential segregation, daytime segregation analysis
 G2020/13 A. BAUER - J. BOUSSARD

Market Power and Labor Share